HIGHER DIMENSIONAL CATENOID, LIOUVILLE EQUATION AND ALLEN-CAHN EQUATION

OSCAR AGUDELO, MANUEL DEL PINO, AND JUNCHENG WEI

ABSTRACT. We build a family of entire solutions to the Allen-Cahn equation in \mathbb{R}^{N+1} for $N \geq 3$, whose level set approaches the higher dimensional catenoid in a compact region and has two logarithmic ends governed by the solutions to the Liouville equation.

Contents

l.	Introduction	1
2.	General geometrical gackground	5
3.	Approximate nodal set	8
1.	Jacobi Operator	14
5.	Approximation and preliminary discussion	18
3.	Lyapunov reduction Scheme	22
7.	Gluing reduction and solution to the projected problem.	27
3.	The proof of Theorem 2	31
References		35

1. Introduction

In this work we construct entire bounded solutions to the semilinear elliptic equation

$$\Delta u + u(1 - u^2) = 0$$
, in \mathbb{R}^{N+1} (1.1)

for any $N \geq 3$. Equation (1.1) is the prototype equation in the gradient theory of phase transition phenomena, developed by Allen-Cahn in [2]. Equation (1.1) also appears as the limit version of the singularly perturbed equation

$$\varepsilon^2 \Delta v + v(1 - v^2) = 0. \quad \text{in } \Omega$$
 (1.2)

via the rescaling $u(x) := v(\varepsilon x)$ in the expanding domain $\varepsilon^{-1}\Omega$, where $\varepsilon > 0$ is a small parameter and for which solutions correspond to the critical points of the energy functional

$$J_{\varepsilon}(v) := \frac{\varepsilon}{2} \int_{\Omega} |\nabla v|^2 + \frac{1}{4\varepsilon} \int_{\Omega} (1 - v^2)^2.$$

In the typical situation in the phase transition phenonema modeled by equation (1.2), the function v represents the phase of a material placed in the region Ω . There are two stables phases represented by the constant functions $v = \pm 1$ each of which corresponds a single phase material and they are global minimizers of the energy J_{ε} . It is then more interesting to study phases in which two different materials coexist. This is equivalent to study solutions of (1.2) connecting the two stables phases ± 1 . A well known result due to Modica, see [26], states that a family of local minimizers of J_{ε} with energy uniformly bounded must converge in L^1 -sense to a function of the form

$$u_*(x) = \chi_M - \chi_{M^c}$$

where $M \subset \Omega$, χ is the characteristic function of a set and ∂M has minimal perimeter. The intuition behind this result was the keystone in the developments of the Γ -convergence during the 70ths and lead to the discovery of the deep connection between the Allen-Cahn equation and the *Theory of Minimal surfaces*. One of the most important developments regarding this connection and concerning entire solutions of equation (1.1) is the celebrated conjecture due to E.De Giorgi, see [12].

DE GIORGI'S CONJECTURE, 1978: Level sets $[u = \lambda]$ of solutions $u \in L^{\infty}(\mathbb{R}^{N+1})$ to problem (1.1) which are monotone in one direction, must be parallel hyperplanes, at least if $1 \leq N \leq 7$. That is equivalent to saying that udepends only on one variable, i.e for some x_0 and some unit vector n,

$$u(x) = w(t), \quad t = (x - x_0) \cdot n$$
 (1.3)

where w(t) is the solution to the one dimensional Allen-Cahn equation

$$w'' + w(1 - w^2) = 0$$
, in \mathbb{R} , $w(0) = 0$, $w'(t) > 0$, $w(\pm \infty) = \pm 1$. (1.4)

De Giorgi's conjecture is in analogy with *Bernstein problem* which states that minimal hypersurfaces that are graphs of entire functions of N variables must be hyperplanes. *Bernstein Problem* is known to be true up to dimension N = 7, see [6, 3, 11, 19]. Bombieri, DeGiorgi and Giusti in [7] proved that Bernstein's statement is false in dimensions $N \geq 8$, by constructing a minimal graph in \mathbb{R}^8 which is not a hyperplane.

De Giorgi's has been proven to be true for N=1 by Ghoussoub and Gui in [21], for N=2 by Ambrosio and Cabre in [4] and for $3 \le N \le 7$ by Savin in [29], under the additional assumption that

$$\lim_{x_{N+1} \to \pm \infty} u(x', x_{N+1}) = \pm 1.$$

This conjecture was recently proven to be false for $N \geq 8$ by del Pino, Kowalczyk and Wei in [16]. The authors used a large dilation of the Bombieri-De Giorgi-Giusti minimal graph constructed in dimensions N=8 as model for the nodal set of the solution they constructed together with a *Lyapunov-Schmidth Reduction* procedure.

The monotonicity assumption in the De Giorgi's conjecture is related to the stability properties of the solutions of (1.1). In this regard, another type of stable solutions in high dimensions was provided by Pacard and Wei in [30], in which the nodal set of these solutions resembles a large dilation of the Simons Cone.

Stability properties of solutions $u \in L^{\infty}(\mathbb{R}^{N+1})$ to (1.1) are studied through their Morse Index. The Morse Index of u, m(u), is defined as the maximal dimension of a vector space E of compactly supported functions ψ for which the quadratic form

$$Q(\psi, \psi) := \int_{\mathbb{R}^{N+1}} |\nabla \psi|^2 - (1 - 3u^2)\psi^2 < 0, \text{ for } \psi \in E - \{0\}.$$

In 2009 a new family of solutions to the Allen-Cahn equation in \mathbb{R}^3 was found by del Pino, Kowalczyk and Wei in [13], in this case the nodal set of these solutions resembles a large dilation of a complete embedded minimal surface with finite total curvature. Example of this kind of surfaces were found by Costa, Hoffman and Meeks, see [24, 8, 23]. The assumption of finite total curvature implies that this type of surfaces have finite Morse Index, a property that is inherited by these solutions and the two Morse indexes coincide. One of these examples is the catenoidal solution having Morse index one and nodal set diverging at logarithmic rate.

In [1], Agudelo, del Pino and Wei constructed two family of unstable solutions to equation (1.1) in \mathbb{R}^3 both of them with nodal set having multiple connected components diverging at infinity at a logarithmic reate. The first of these families having Morse Index one with nodal set ruled by the *Toda System* in \mathbb{R}^2 and the other one having large Morse index and with nodal set resembling, far away from a compact set, a set of nested catenoids, having therefore a contrasting nature respect to the examples found in [13].

All the developments mentioned above show the strong connection between the Allen-Cahn equation and the Minimal surfaces theory, but this connection has been only partly explored, in particular when providing more examples of solutions to the Allen-Cahn Equation in high dimensions. On the other hand, unlike dimension \mathbb{R}^3 in which a large amount of examples of minimal surfaces exist and have been analyzed, in higher dimensions \mathbb{R}^{N+1} , $N \geq 3$, there are very few examples of minimal surfaces. The catenoid being among the classical ones.

In this paper we explore the connection between the higher dimensional catenoid and the equation (1.1) for $N \geq 3$. To state our result we consider M to be the N-dimensional catenoid, which is described by the graph of the axially symmetric functions $\pm F$ where F = F(|y|) is the unique increasing axially symmetric solution to the minimal surface equation for graphs

$$\nabla \cdot \left(\frac{\nabla F}{\sqrt{1+|\nabla F|^2}}\right) = 0, \quad |y| > 1, \quad y \in \mathbb{R}^N$$
(1.5)

with the initial conditions

$$F(1) = 0, \quad \partial_r F(1) = +\infty.$$

The catenoid M has asymptotically parallel flat ends, a fact that is reflected in the asymptotics of the function F

$$F(r) = T - \frac{r^{2-N}}{N-2} + \mathcal{O}_{L^{\infty}(\mathbb{R}^N)} \left(r^{4-3N} \right), \quad \text{as } r \to \infty$$
 (1.6)

with

$$T := \int_{1}^{\infty} \frac{1}{\sqrt{s^{2(N-1)} - 1}} ds$$

and this relations in (3.2) can be differentiated, see [18] and references therein.

At a first glance, one may think that the proper choice for the approximate nodal set of the solutions predicted in our theorem would be a large dilated version of M, $M_{\varepsilon} = \varepsilon^{-1}M$, with $\varepsilon > 0$ small. As pointed out in [13] and in section 7 of [20], this is not an appropriate global choice. Instead, the parallel ends of M_{ε} must be perturbed in order to obtain a profile for the nodal set that will lead to good sizes in the error.

The rule governing this perturbation is the Liouville Equation

$$\varepsilon \Delta F_{\varepsilon} - a_0 \, e^{\frac{-2\sqrt{2} \, F_{\varepsilon}}{\varepsilon}} = 0, \quad \text{for } |y| > R_{\varepsilon}$$
 (1.7)

where for some $\bar{a} > 0$

$$R_{\varepsilon} \approx \bar{a} \, \varepsilon^{-1} e^{\frac{\sqrt{2}T}{\varepsilon}}, \quad \text{as } \varepsilon \to 0.$$

The constant $a_0 > 0$ is given by

$$a_0 = \|w'\|_{L^2(\mathbb{R})}^{-2} \int_{\mathbb{R}} 6(1 - w^2(t))e^{\sqrt{2}t}w'(t) dt > 0$$

where the function w(t) is the solution of (1.4), which is given explicitly by $w(t) = \tanh(t/\sqrt{2})$.

We match the functions F and F_{ε} in a C^1 way by considering the additional initial conditions for F_{ε} at $r = R_{\varepsilon}$

$$F_{\varepsilon}(R_{\varepsilon}) = T - \frac{R_{\varepsilon}^{2-N}}{N-2} + \mathcal{O}\left(R_{\varepsilon}^{4-3N}\right), \quad \partial_{r}F_{\varepsilon}(R_{\varepsilon}) = R_{\varepsilon}^{1-N} + \mathcal{O}\left(R_{\varepsilon}^{3(1-N)}\right)$$

from where it is possible to conclude that as $r \to \infty$

$$F_{\varepsilon}(r) = \frac{\varepsilon}{2\sqrt{2}} \left(\log \left(\frac{2\sqrt{2} a_0}{\varepsilon^2} \right) + 2\log(r) - \log(2(N-2)) + \bar{w}_{\varepsilon}(r) \right)$$
(1.8)

where the function $\bar{w}_{\varepsilon}(r)$ satisfies for $r > R_{\varepsilon}$ that

$$|r \,\partial_r \bar{w}_{\varepsilon}(r)| + |\bar{w}_{\varepsilon}(r)| \le \begin{cases} C\left(\frac{r}{R_{\varepsilon}}\right)^{-\frac{N-2}{2}}, & 3 \le N \le 9\\ C\log\left(\frac{r}{R_{\varepsilon}}\right)\left(\frac{r}{R_{\varepsilon}}\right)^{-4}, & N = 10\\ C\left(\frac{r}{R_{\varepsilon}}\right)^{\lambda_+}, & N \ge 11 \end{cases}$$

$$(1.9)$$

where $\lambda_{+} > 0$ is defined at (3.9) below.

Denote

$$x = (x', x_{N+1}) \in \mathbb{R}^N \times \mathbb{R}, \quad r(x) := |x'|,$$

and set

$$G(x') = (1 - \chi_{\varepsilon}(|x'|))F(x') + \chi_{\varepsilon}(|x'|)F_{\varepsilon}(\varepsilon x'), \quad |x'| > 1$$

where χ_{ε} is the characteristic function of the set $\mathbb{R}^N - B_{R_{\varepsilon}}$. Define Σ to be the revolution hypersurface, resulting from the union of the graphs of $\pm G$ and Σ_{ε} a ε^{-1} dilation of Σ . The hypersurface Σ_{ε} is a connected complete embedded orientable surface, splitting \mathbb{R}^{N+1} into two connected components. We will prove the following theorem.

Theorem 1. For every $N \geq 3$ and any sufficiently small $\varepsilon > 0$ there exist a solution u_{ε} to equation (1.1) having the asymptotics

$$u_{\varepsilon}(x) = w(z)(1 + o(1)), \quad as \ \varepsilon \to 0$$

uniformly on every compact subset of \mathbb{R}^N , where z is the normal direction to the largely dilated surface Σ_{ε} described above and w(t) is the heteroclinic solution to (1.4) connecting ± 1 .

Even more, as for every $\varepsilon > 0$ small and $x_{N+1} \in \mathbb{R}$, as $|x'| \to \infty$

$$u_{\varepsilon}(x) = w\left(x_{N+1} - \varepsilon^{-1} F_{\varepsilon}(\varepsilon x')\right) - w\left(\varepsilon^{-1} F_{\varepsilon}(\varepsilon x') - x_{N+1}\right) + 1 + o(1), \qquad x = (x', x_{N+1})$$
where $F_{\varepsilon}(x')$ is the radially symmetric function described in (1.8).

As we will see throughout the proof, we will be able to give a more precise description of these solutions and their nodal sets, not only as $\varepsilon \to 0$, but also at infinity.

The strategy to prove Theorem 1 consist in apply the infinite dimensional reduction method in the spirit of the works [1, 13, 16]. One of the crucial steps in this method is the choice of an accurate approximation of a solution to (1.1). In this regard, and as pointed out by Modica, it is fundamental to select properly the set where the solution is expected to change sign.

The first candidate for nodal set would be a small perturbation of the large dilation of M, $\varepsilon^{-1}M$. Since $\varepsilon^{-1}M$ has parallel ends, the approximation in the upper end will take the approximate form

$$w\left(\frac{T}{\varepsilon} + x_{N+1} + h(\varepsilon x')\right) - w\left(\frac{T}{\varepsilon} - x_{N+1} - h(\varepsilon x')\right) + 1, \quad \text{for } x = (x', x_{N+1}), \quad |x'| \to \infty, \quad x_{N+1} \in \mathbb{R}.$$

Setting $t = \frac{T}{\varepsilon} + x_{N+1} + h(\varepsilon x')$, the approximation can be written as

$$u := w(t) - w(2\varepsilon^{-1}T - 2h(\varepsilon x') + t) - 1$$

and we find that

$$u(1-u^2) \approx 6(1-w^2(t))e^{-\sqrt{2}t}e^{\frac{-2\sqrt{2}T}{\varepsilon}}$$
.

This last expression states that the Allen-Cahn nonlinearity takes account for the interaction of parallel ends.

It follows that at main order the error created by this approximation reads as

$$-\varepsilon^2 \Delta_{\mathbb{R}^N} h \, w'(t) + 6(1 - w^2(t)) e^{-\sqrt{2}t} e^{\frac{-2\sqrt{2}T}{\varepsilon}}$$

and after the reduction procedure, the reduced equation to adjust the nodal set would read as

$$\Delta_{\mathbb{R}^N} h \sim a_0 \varepsilon^{-2} e^{\frac{-2\sqrt{2}T}{\varepsilon}}$$

where the right hand side is bounded, small but has no decay. Therefore

$$h(x') \sim a_0 e^{-\sqrt{2}t} e^{\frac{-2\sqrt{2}T}{\varepsilon}} |x'|^2, \qquad |x'| \to \infty$$

which leads to a nodal set that diverges at a fast rate away from the catenoid $\varepsilon^{-1}M$, as $|x'| \to \infty$. We solve this issue by using instead of $\varepsilon^{-1}M$, the hypersurface $\varepsilon^{-1}\Sigma$.

Our second result shows a dramatic difference between the Allen-Cahn equation and the Theory of Minimal surfaces. A result proven in [28] states that for dimensions $3 \le N \le 9$, the catenoid M has Morse index 1. As treated in [13], the Morse Index of the catenoid is defined as the number of negative eigenvalues of the linearization of the mean curvature operator associated to M. In contrast our result states for $\varepsilon > 0$ small and with the same restriction in the dimension, these solutions are highly unstable.

Theorem 2. Let u_{ε} be the solution to equation (1.1) constructed in Theorem 1. Then for $\varepsilon > 0$ small and for dimensions $3 \leq N \leq 9$, $m(u_{\varepsilon}) = +\infty$.

As we will see from the proof of Theorem 2, in low dimensions the infinite Morse Index, comes from the asymptoics described in (1.10) and the Morse Index of the function F_{ε} which comes into play far away. In dimensions $N \geq 10$, due to Hardy's inequality, we expect for these solutions to have Morse Index 1, the Morse Index of M. This, in view of the results in [1, 13] and since there would be no negative directions related to the function F_{ε} .

The paper is structured as follows: Section 2 sets up the general geometric setting need for the proof of Theorem 1. Section 3 describes the asymptotics of the hypersurface that will be the model for the nodal set of the solutions predicted in our result while section 4 deals with the solvability theory for a Jacobi-Hardy type of operator related to the reduced problem. Section 6 describes the strategy and details of the proof of the Theorem 1 which, as mentioned before, relies on an infinite dimensional reduction procedure, a method motivated by the pioneering work of [20]. Section 7 contains some technical results used in section 6 and finally the detailed proof of Theorem 2 is provided in the last section.

Acknowledgments: O. Agudelo has been supported by the Grant 13- 00863S of the Grant Agency of the Czech Republic. M. del Pino has been supported by grants Fondecyt 1150066, Fondo Basal CMM and Millenium Nucleus CAPDE NC130017. J. Wei is supported by a NSERC grant from Canada.

2. General geometrical gackground

Throughout our discussion we adopt the following conventions. We write $\mathbb{R}^{N+1} = \mathbb{R}^N \times \mathbb{R}$ and we denote

$$x = (x', x_{N+1}) \in \mathbb{R}^{N+1}, \quad r(x', x_{N+1}) := |x'|.$$

Let Σ be a smooth orientable N- dimensional embedded hypersurface in \mathbb{R}^{N+1} . Next, we describe the euclidean Laplacian close to the large dilation of Σ , $\varepsilon^{-1}\Sigma$, in a system of coordinates that is suitable for the developments in subsequent sections.

For any point $y \in \Sigma$, we write

$$y = (y', y_{N+1}) \in \Sigma \subset \mathbb{R}^{N+1}$$

and we denote by $\nu: \Sigma \to S^N$ a fixed choice of the unit normal vector to the surface Σ .

It is easy to check that if $Y: \mathcal{U} \subset \mathbb{R}^N \to \mathbb{R}^{N+1}$ is a local parametrization of the surface Σ with

$$s = (s_1 \dots, s_N) \in \mathcal{U} \to y = Y(s) \in \Sigma$$

then the map

$$X(s,z) = Y(s) + z \nu(s)$$

provides local coordinates in a neighborhood $\mathcal N$ of the form

$$\mathcal{N} := \{ x = X(s, z) : |z| < \delta_0 \}$$

for some fixed $\delta_0 > 0$.

System of coordinates X(s, z) is known as the Fermi coordinates associated to Σ . Observe that for a point $x = X(s, z) \in \mathcal{N}$, the variable z represents the signed distance to Σ , i.e

$$|z| = \operatorname{dist}(\Sigma, x).$$

For z small and fixed such that

$$X(\mathcal{U},z)\subset\mathcal{N}$$

we denote by Σ_z the translated surface locally parameterized by

$$X(\cdot, z): U \to X(\mathcal{U}, z), \qquad s \in \mathcal{U} \to X(s, z) \in \mathcal{N}.$$

We denote by $g := (g_{ij})_{N \times N}$ and $g_z := (g_{z,ij})_{N \times N}$ the metrics on Σ and Σ_z respectively, with inverses $g^{-1} := (g^{ij})_{N \times N}$ and $g_z^{-1} := (g_z^{ij})_{N \times N}$. We find that

$$g_{z,ij} = g_{ij} + z\hat{a}_{ij}(s) + z^2 \hat{b}_{ij}(s), \quad s \in \mathcal{U}, \quad |z| < \delta_0$$
 (2.1)

where for $i, j = s_1, \ldots, s_N$

$$g_{ij} = \langle \partial_i Y; \partial_j Y \rangle$$

and

$$\hat{a}_{ij} = \langle \partial_i Y; \partial_j \nu \rangle + \langle \partial_j Y; \partial_i \nu \rangle, \quad \hat{b}_{ij} = \langle \partial_i \nu; \partial_j \nu \rangle.$$

Hence, we may write

$$g_z = g + z A(s) + z^2 B(s)$$

where $A = (\hat{a}_{ij})_{N \times N}$, $B = (\hat{b}_{ij})_{N \times N}$.

It is well-known that in \mathcal{N} , we can compute the euclidean Laplacian in Fermi coordinates using the formula

$$\Delta = \partial_z^2 + \Delta_{\Sigma_z} - H_{\Sigma_z} \partial_z$$

where Δ_{Σ_z} is the Laplace-Beltrami operator of Σ_z and H_{Σ_z} its mean curvature.

The expression for Δ_{Σ_z} is given by

$$\Delta_{\Sigma_{z}} = \frac{1}{\sqrt{\det g_{z}}} \partial_{i} \left(\sqrt{\det g_{\Sigma_{z}}} g_{z}^{ij} \partial_{j} \right)
= g_{z}^{ij} \partial_{ij} + \frac{1}{\sqrt{\det g_{z}}} \partial_{i} \left(\sqrt{\det g_{z}} g_{z}^{ij} \right) \partial_{j}
= g_{z}^{ij} \partial_{ij} - g_{z}^{kl} \Gamma_{z,kl}^{i} \partial_{i}$$
(2.2)

and $\Gamma_{z,kl}^i$ stand for the Christoffel symbols on Σ_z and where in every term, summation is understood over repeated indexes.

Using (2.1) and (2.2), we write

$$\Delta_{\Sigma_z} = \Delta_{\Sigma} + \mathbb{A}_{ij}\partial_{ij} + \mathbb{B}_i\partial_i$$

where

$$\begin{array}{lcl} \mathbb{A}_{ij} & = & g_z^{ij} - g^{ij} \\ \mathbb{B}_i & = & g_z^{kl} \left[\Gamma_{z,kl}^i - \Gamma_{kl}^i \right] + \Gamma_{kl}^i \left[g_z^{kl} - g^{kl} \right]. \end{array}$$

From expression (2.1) it is direct to verify that for some smooth bounded functions $a_{ij(s,z)}$ and $b_i(s,z)$

$$\mathbb{A}_{ij}(s,z) = z \, a_{ij}(s,z), \quad \mathbb{B}_i(s,z) = z \, b_i(s,z), \quad \text{in } \mathcal{N}. \tag{2.3}$$

In further developments, the geometrical setting will become more specific, allowing us to provide more precise information on the asymptotic behavior of these functions.

Next, we compute the mean curvature of Σ_z , H_{Σ_z} . Denote by κ_i the principal curvatures of Σ . From the well known formula

$$H_{\Sigma_z} = \sum_{i=1}^N \frac{\kappa_i}{1 - z \,\kappa_i} = \sum_{i=1}^N \kappa_i + z \kappa_i^2 + z^2 \kappa_i^3 + \cdots$$

from where we obtain

$$H_{\Sigma_z} = H_{\Sigma} + z |A_{\Sigma}|^2 + z^2 b_{N+1}(s, z)$$
 (2.4)

with

$$b_{N+1}(s,z) = \sum_{i=1}^{N} \kappa_i^3 + z \kappa_i^4 + \dots$$

and H_{Σ} and $|A_{\Sigma}|$ denote respectively the mean curvature and the norm of the second fundamental form of the hypersurface Σ .

Summarizing, we have the formula for the euclidean Laplacian expressed in Fermi coordinates in \mathcal{N}

$$\Delta_X = \partial_{zz} + \Delta_{\Sigma} - (H_{\Sigma} + z|A_{\Sigma}|^2) \partial_z + + z a_{ij}(s, z) \partial_{ij} + z b_i(s, z) \partial_i - z^2 b_{N+1}(s, z) \partial_z.$$
(2.5)

Next, let us consider the dilated surface $\Sigma_{\varepsilon} := \varepsilon^{-1}\Sigma$ with local coordinate system given naturally by $Y_{\varepsilon} : \mathcal{U}_{\varepsilon} \to \Sigma_{\varepsilon}$

$$Y_{\varepsilon}(s) = \varepsilon^{-1}Y(\varepsilon s), \quad s \in \mathcal{U}_{\varepsilon} := \varepsilon^{-1}\mathcal{U}.$$

It is direct to check that the unit normal vector to Σ_{ε} is given by

$$\nu_{\varepsilon}(y) = \nu(\varepsilon y), \quad y \in \Sigma_{\varepsilon}$$

so that the dilated Fermi coordinates read as

$$X_{\varepsilon}(s, z) = Y_{\varepsilon}(s) + z\nu_{\varepsilon}(s)$$

= $\varepsilon^{-1}Y(\varepsilon s) + z\nu(\varepsilon s)$

which provide local coordinates in the dilated neighborhood

$$\mathcal{N}_{\varepsilon} := \{ x = X_{\varepsilon}(s, z) : |\varepsilon z| \le \delta_0, \quad s \in \bar{\mathcal{U}}_{\varepsilon} \}$$

We also readily check that

$$\Delta_{X_{\varepsilon}} = \varepsilon^2 \Delta_X$$

so that, rescaling formula (2.5) we obtain

$$\Delta_{X_{\varepsilon}} = \partial_{zz} + \Delta_{\Sigma_{\varepsilon}} - (\varepsilon H_{\Sigma} + \varepsilon^{2} z |A_{\Sigma}|^{2}) \partial_{z} + \varepsilon z a_{ij}(\varepsilon s, \varepsilon z) \partial_{ij} + \varepsilon^{2} b_{i}(\varepsilon s, \varepsilon z) \partial_{i} + \varepsilon^{3} z^{2} b_{N+1}(\varepsilon s, \varepsilon z) \partial_{z}. \quad (2.6)$$

The previous computations are not enough for the proof of our results. We need to introduce a smooth bounded parameter function h defined in Σ . Abusing the notation we set in the coordinates $Y: \mathcal{U} \to \Sigma$

$$h(Y(s)) = h(s), \quad s \in \bar{\mathcal{U}}$$

and we consider the translated coordinate system

$$X_{\varepsilon,h}(s,t) = X_{\varepsilon}(s,t+h(\varepsilon s))$$

and the translated and dilated neighborhood

$$\mathcal{N}_{\varepsilon,h} := \left\{ x = X_{\varepsilon,h}(s,t) : |t + h(\varepsilon s)| < \frac{\delta_0}{\varepsilon} \right\}$$

Then, by setting $z = t + h(\varepsilon s)$, we can compute the euclidean Laplacian in the translated Fermi coordinates as in [1, 13], i.e

$$\Delta_{X_{\varepsilon,h}} = \partial_{tt} + \Delta_{\Sigma_{\varepsilon}} - \varepsilon H_{\Sigma} \partial_{t} - \varepsilon^{2} |A_{\Sigma}|^{2} t \partial_{t} - \varepsilon^{2} \{\Delta_{\Sigma} h + |A_{\Sigma}|^{2} h\} \partial_{t} - \varepsilon \partial_{i} h \partial_{ti} + \varepsilon^{2} \partial_{i} h^{2} \partial_{tt} + D_{\varepsilon,h} \quad (2.7)$$

where

$$D_{\varepsilon,h} = \varepsilon (t+h)a_{ij}(\varepsilon s, \varepsilon(t+h)) \left\{ \partial_{ij} - \varepsilon \partial_i h \, \partial_{tj} - \varepsilon^2 \, \partial_{ij} h \, \partial_t + \varepsilon^2 \, \partial_i h^2 \, \partial_{tt} \right\}$$
$$+ \varepsilon^2 (t+h) b_i(\varepsilon s, \varepsilon(t+h)) \left\{ \partial_i - \varepsilon \, \partial_i h \, \partial_t \right\} + \varepsilon^3 (t+h)^2 b_{N+1} \, \partial_t$$
 (2.8)

and where the functions H_{Σ} , $|A_{\Sigma}|^2$, h, $\partial_i h$, $\partial_{ij} h$ are evaluated at εs .

3. Approximate nodal set

In this section we describe the asymptotics of the nodal set for the family of solutions predicted in Theorem 1 and we compute some geometric quantities which are crucial for our developments.

Let F = F(|y|) be the unique increasing axially symmetric solution to the minimal surface equation for graphs

$$\nabla \cdot \left(\frac{\nabla F}{\sqrt{1+|\nabla F|^2}}\right) = 0, \quad |y| > 1, \quad y \in \mathbb{R}^N$$
(3.1)

with the initial conditions

$$F(1) = 0, \quad \partial_r F(1) = +\infty.$$

From the symmetry of F, equation (3.1) can be written as the ODE

$$r^{N-1}\partial_r F - (1+|\partial_r F|^2)^{\frac{1}{2}} = 0, \quad r > 1$$

from where we directly obtain that F(r) is strictly increasing and F(r) > 0 for every r > 1. We also obtain the asymptotics

$$F(r) = \begin{cases} \sqrt{\frac{2(r-1)}{N-1}} \left(1 + \mathcal{O}_{L^{\infty}(\mathbb{R}^{N})}(r-1) \right), & \text{as } r \to 1 \\ T - \frac{r^{2-N}}{N-2} + \mathcal{O}_{L^{\infty}(\mathbb{R}^{N})}\left(r^{4-3N}\right), & \text{as } r \to \infty \end{cases}$$
(3.2)

with

$$T := \int_{1}^{\infty} \frac{1}{\sqrt{s^{2(N-1)} - 1}} \, ds$$

and relations in (3.2) can be differentiated for r > 1.

Consider the smooth hypersurfaces M_{\pm} , parameterized by

$$y \in D \mapsto (y, \pm F(y)) \in \mathbb{R}^N \times \mathbb{R}$$

respectively and where $D := \mathbb{R}^N - B_1$.

The hypersurface $M = M_+ \cup M_-$ is the N-dimensional catenoid and from (3.1) it is a minimal surface of revolution. An important consequence from (3.2) is that M has two almost flat parallel ends.

As mentioned above and as pointed out in [13] and in section 7 of [20], a large dilated version of M, $M_{\varepsilon} = \varepsilon^{-1}M$, with $\varepsilon > 0$ small must be perturbed in order to obtain the right profile for the nodal set that will lead to good sizes in the error.

To do so, we consider $\varepsilon > 0$ small and fixed and we also fix a large radius R_{ε} to be chosen appropriately. Next step consists in fixing a particular *smooth radial solution* F_{ε} to the Liouville equation

$$\varepsilon \Delta F_{\varepsilon} - a_0 \, e^{\frac{-2\sqrt{2} \, F_{\varepsilon}}{\varepsilon}} = 0, \quad \text{for } |y| > R_{\varepsilon}$$
 (3.3)

where

$$a_0 := \|w'\|_{L^2(\mathbb{R})}^{-2} \int_{\mathbb{R}} 6(1 - w^2(t)) e^{\sqrt{2}t} w'(t) dt > 0.$$

It is enough to match in a C^1 topology the catenoid M with the graphs of F_{ε} and $-F_{\varepsilon}$ outside of a large compact subset of M, determined by the large radius R_{ε} . Taking into account the asymptotics in (3.2) for M, we consider (3.3) with initial conditions for F_{ε} at $r = R_{\varepsilon}$, namely

$$F_{\varepsilon}(R_{\varepsilon}) = T - \frac{R_{\varepsilon}^{2-N}}{N-2} + \mathcal{O}\left(R_{\varepsilon}^{4-3N}\right)$$

$$\partial_r F_{\varepsilon}(R_{\varepsilon}) = R_{\varepsilon}^{1-N} + \mathcal{O}\left(R_{\varepsilon}^{3(1-N)}\right).$$

The size of the radius R_{ε} and the asymptotic behavior of the function F_{ε} , both in terms of ε , can be found by using the scaling

$$F_{\varepsilon}(y) = \frac{\varepsilon}{2\sqrt{2}} \left[\log \left(\frac{2\sqrt{2} a_0 R_{\varepsilon}^2}{\varepsilon^2} \right) + f\left(\frac{|y|}{R_{\varepsilon}} \right) \right]$$

so that

$$\Delta f - e^{-f} = 0, \quad |y| > 1, \quad y \in \mathbb{R}^N.$$

The initial conditions for F_{ε} translate into the initial conditions for f

$$f(1) = \frac{2\sqrt{2}T}{\varepsilon} - \log\left(\frac{2\sqrt{2}a_0R_{\varepsilon}^2}{\varepsilon^2}\right) - \frac{2\sqrt{2}a_0R_{\varepsilon}^{2-N}}{\varepsilon(N-2)} + \mathcal{O}\left(R_{\varepsilon}^{4-3N}\varepsilon^{-1}\right)$$

$$\partial_r f(1) = 2\sqrt{2} \varepsilon^{-1} \left(R_\varepsilon^{2-N} + \mathcal{O}\left(R_\varepsilon^{4-3N} \right) \right).$$

Using the dependence on $\varepsilon > 0$ of the initial conditions, the Intermediate Value Theorem and the Implicit Function Theorem, for every $\varepsilon > 0$ small there exists a radius R_{ε} with asymptotics described as

$$R_{\varepsilon} = \bar{a} \, \varepsilon \, e^{\frac{\sqrt{2}T}{\varepsilon}} \left(1 + \mathcal{O}\left(\varepsilon^{1-N} e^{-\frac{\sqrt{2}(N-2)T}{\varepsilon}}\right) \right), \quad \text{as } \varepsilon \to 0$$
(3.4)

for some positive constant $\bar{a} > 0$ and such that f(1) = 0. Consequently we find that

$$\partial_r f(1) = \varepsilon^{-1} \left(R_{\varepsilon}^{2-N} + \mathcal{O} \left(R_{\varepsilon}^{4-3N} \right) \right)$$

and we directly check that

$$F_{\varepsilon}(y) = T - \frac{R_{\varepsilon}^{2-N}}{N-2} + \mathcal{O}\left(R_{\varepsilon}^{4-3N}\right) + \frac{\varepsilon}{2\sqrt{2}} f\left(\frac{|y|}{R_{\varepsilon}}\right).$$

Thus, it remains to analyze the asymptotic behavior of solutions to the problem

$$\Delta f - e^{-f} = 0$$
, for $|y| > 1$, $f(1) = \bar{f}$, $\partial_r f(1) = \delta$ (3.5)

in the class of radially symmetric functions and where $\delta > 0$ is small.

We pass to the Emden-Fowler variable $s = \log(r)$ and we set f(r) = v(s) to find that

$$v''(s) + (N-2)v'(s) - e^{2s-v(s)} = 0, \quad s > 0, \quad v(0) = \bar{f}, \quad v'(0) = \delta.$$
(3.6)

Classical ODE theory immplies that the initial value problem has always a solution in a maximal interval $(0, S_*)$. Let us now assume that

$$v(s) = 2s + \log\left(\frac{1}{2(N-2)}\right) + w(s), \quad 0 < s < S_*$$

so that

$$w''(s) + (N-2)w'(s) - 2(N-2)\left[e^{-w(s)} - 1\right] = 0, \quad 0 < s < S_*$$
(3.7)

$$w(0) = \bar{f} - \log\left(\frac{1}{2(N-2)}\right), \quad w'(0) = -(2-\delta).$$

Considering the Lyapunov functional

$$L(\mathbf{s}) = \frac{1}{2}|w'(\mathbf{s})|^2 + 2(N-2)e^{-w(\mathbf{s})} + 2(N-2)w(\mathbf{s}), \quad \mathbf{s} \in (0, S_*)$$

we find that

$$\frac{d}{ds}L(s) = -(N-2)|w'(s)|^2 \le 0$$

and since $N \geq 3$, it follows that

$$L(s) \le L(0) \le C$$

where C > 0 does not depend on $\delta > 0$ small. This implies that w(s) and w'(s) remain uniformly bounded and so they are defined for every s > 0.

About the asymptotics of w(s), we proceed as in section 2 in [10]. Observe that we can write equation (3.7) as

$$w'' + (N-2)w' + 2(N-2)w - N(w) = 0, \quad s > 0$$
(3.8)

$$N(w) = 2(N-2) [e^{-w} - 1 + w].$$

The indicial roots associated to the linear part in (3.8) are the solutions to the algebraic equation

$$\lambda^{2} + (N-2)\lambda + 2(N-2) = 0$$

from where

$$\lambda_{\pm} = -\frac{N-2}{2} \pm \frac{1}{2} \sqrt{(N-2)(N-10)}.$$
 (3.9)

Hence, a fundamental set for the linear part of equation (3.8) is given by

$$w_{+}(\mathbf{s}) := e^{-\frac{N-2}{2}} \mathbf{s} \cdot \begin{cases} \cos\left(\frac{1}{2}\sqrt{(N-2)(10-N)}\,\mathbf{s}\right), & 3 \le N \le 9 \\ 1, & N = 10 \end{cases}$$

$$e^{+\frac{1}{2}\sqrt{(N-2)(N-10)}\,\mathbf{s}}, & N \ge 11$$

$$(3.10)$$

$$w_{-}(s) = e^{-\frac{N-2}{2}s} \cdot \begin{cases} \sin\left(\frac{1}{2}\sqrt{(N-2)(10-N)}s\right), & 3 \le N \le 9 \\ s, & N = 10 \end{cases}$$

$$e^{-\frac{1}{2}\sqrt{(N-2)(N-10)}s}, & N \ge 11$$

$$(3.11)$$

with Wronski determinant

$$W(s) := \begin{cases} \frac{1}{2}\sqrt{(N-2)(10-N)}e^{-\frac{N-2}{2}s}, & 3 \le N \le 9 \\ e^{-8s}, & N = 10 \\ \frac{1}{2}\sqrt{(N-2)(N-10)}e^{-\frac{N-2}{2}s}, & N \ge 11. \end{cases}$$
(3.12)

Therefore, we can write

$$w(s) = A w_{+}(s) + B w_{-}(s) + \bar{w}(s)$$

where \bar{w} is a particular solution to the equation

$$\bar{w}'' + (N-2)\bar{w}' + 2(N-2)\bar{w} = -N(w), \quad s > 0.$$

We only describe the resonant case N=10, since the other cases can be described in a similar fashion. Using variations of parameters formula when N=10, we choose \bar{w} to be the function

$$\bar{w}(s) = e^{-4s} \int_{s}^{\infty} \tau e^{4\tau} N(w) d\tau - s e^{-4s} \int_{s}^{\infty} e^{4\tau} N(w) d\tau.$$

Using a fixed point argument, the following Lemma readily follows.

Lemma 3.1. The function v solving equation (3.6) has the following asymptotic behavior

$$v(s) = 2s + \log\left(\frac{1}{2(N-2)}\right) + w(s), \quad 0 < s < \infty$$

where the function w(s) satisfies, as $s \to \infty$

$$|w(s)| \le \begin{cases} Ce^{-\frac{N-2}{2}s}, & 3 \le N \le 9\\ Cse^{-4s}, & N = 10\\ Ce^{\lambda+s}, & N \ge 11 \end{cases}$$
 (3.13)

where

$$\lambda_{+} = -\frac{N-2}{2} + \frac{1}{2}\sqrt{(N-2)(N-10)} < 0.$$

and these relations can be differentiated in s > 0.

Lemma 3.1 can be restated for the function $f(r) = v(\log(r))$ as follows.

Corollary 3.1. The function f solving problem (3.5) has the asymptotics as $r \to \infty$

$$f(r) = 2\log(r) - \log(2(N-2)) + \bar{\omega}(r)$$

where

$$|r \,\partial_r \,\bar{\omega}(r)| + |\bar{\omega}(r)| \le \begin{cases} C \, r^{-\frac{N-2}{2}}, & 3 \le N \le 9\\ C \, \log(r) \, r^{-4}, & N = 10\\ C \, r^{\lambda_+}, & N \ge 11 \end{cases}$$
(3.14)

Summarizing, for every $\varepsilon > 0$ small

$$F_{\varepsilon}(r) = T - \frac{R_{\varepsilon}^{2-N}}{N-2} + \mathcal{O}\left(R_{\varepsilon}^{4-3N}\right) + \frac{\varepsilon}{2\sqrt{2}} f\left(\frac{|y|}{R_{\varepsilon}}\right), \quad r > R_{\varepsilon}$$

and as $r \to \infty$

$$F_{\varepsilon}(r) = \frac{\varepsilon}{2\sqrt{2}} \left(\log \left(\frac{2\sqrt{2} a_0}{\varepsilon^2} \right) + 2\log(r) - \log(2(N-2)) + \bar{\omega}_{\varepsilon}(r) \right). \tag{3.15}$$

where the function $\bar{\omega}_{\varepsilon}(r)$ satisfies for $r > R_{\varepsilon}$ that

$$|r \,\partial_r \bar{\omega}_{\varepsilon}(r)| + |\bar{\omega}_{\varepsilon}(r)| \le \begin{cases} C\left(\frac{r}{R_{\varepsilon}}\right)^{-\frac{N-2}{2}}, & 3 \le N \le 9\\ C\log\left(\frac{r}{R_{\varepsilon}}\right)\left(\frac{r}{R_{\varepsilon}}\right)^{-4}, & N = 10\\ C\left(\frac{r}{R_{\varepsilon}}\right)^{\lambda_+}, & N \ge 11 \end{cases}$$
(3.16)

these relations can be differentiated in $r > R_{\varepsilon}$ and where we recall from (3.4) that

$$R_{\varepsilon} \approx \bar{a} \, \varepsilon^{-1} e^{\frac{\sqrt{2}T}{\varepsilon}}, \quad \text{as } \varepsilon \to 0.$$

Consider now a characteristic function $\chi: \mathbb{R}_+ \to [0,1]$ such that

$$\chi(s) = \begin{cases} 0, & s < 1\\ 1, & s > 1 \end{cases}$$

and denote $\chi_{\varepsilon}(r) = \chi(\frac{r}{R_{\varepsilon}})$.

As predicted abov, we glue the catenoid M with the graphs of $\pm F_{\varepsilon}$ by means of the axially symmetric function G(r) described by

$$G(r) = (1 - \chi_{\varepsilon}) F(r) + \chi_{\varepsilon} F_{\varepsilon}(r), \quad r > 1.$$
(3.17)

Let Σ_+ be the graph of the function +G described by

$$\Sigma_{+} = \{(x', x_{N+1}) : x_{N+1} = +G(x'), |x'| > 1\}.$$

Analogously we define Σ_{-} as the graph of -G. Define also

$$\Sigma := \Sigma_+ \cup \Sigma_-.$$

For notational simplicity we have omitted the explicit dependence on $\varepsilon > 0$ of G and Σ . Notice also that $\bar{\Sigma}$ is a C^1 connected hypersurface of revolution having the x_{N+1} -axis as its axis of symmetry and dividing \mathbb{R}^{N+1} into two connected components say S_+ and S_- , where we choose S_+ to be the component containing the axis of symmetry.

We consider the local parametrization for Σ_{\pm} in polar coordinates

$$Y_{\pm}(r,\Theta) = (r\Theta, \pm G(r)), \quad Y_{\pm} : (1,+\infty) \times S^{N-1} \to \mathbb{R}^{N+1}$$

Abusing the notation, we know that the normal vector to Σ_{\pm} , $\nu_{\pm}: \Sigma_{\pm} \to S^N$ pointing towards the x_{N+1} -axis and in terms of the local coordinates (r, Θ) is given by

$$\nu_{\pm}(r,\Theta) = \frac{\pm 1}{\sqrt{1 + |\partial_r G(r)|^2}} \left(-\partial_r G(r)\Theta, 1\right), \quad r > 1$$

respectively in Σ_{\pm} and we have associated Fermi coordinates

$$X_{\pm}(r,\Theta,z) = Y_{\pm}(r,\Theta) + z \nu_{\pm}(r,\Theta)$$

for r > 1 and $\Theta \in S^N$.

It is not hard to check using (3.2) and (3.15) that X_{\pm} provide local diffeomorphism in a neighborhood of the form

$$\mathcal{N}_{\pm} := \left\{ X_{\pm}(r,\Theta,z) : |z| \le \delta_0 \ (1 - \eta_{\varepsilon}) + \eta_{\varepsilon} F_{\varepsilon}(r), \quad r > 1, \quad \Theta \in S^{N-1} \right\}$$

where the constant $\delta_0 = \delta_0(M) > 0$ depends only on the catenoid M.

Also we remark that the function G is not only C^1 , but rather C^2 except at the point $r = R_{\varepsilon}$. From this, it is not difficult to verify that formula (2.5) also holds almost everywhere in this neighborhood.

Since, we are working in an axially symmetric setting we know that all the geometric quantities mentioned in section 2 share also this symmetry. In particular we know that

$$\Delta_{\Sigma_{\pm}} = \frac{1}{r^{N-1}\sqrt{1+|\partial_r G(r)|^2}} \partial_r \left(\frac{r^{N-1}}{\sqrt{1+|\partial_r G(r)|^2}} \partial_r \right)$$

$$= \frac{\partial_{rr}}{1+|\partial_r G(r)|^2} + \left(\frac{N-1}{r} - \frac{\partial_r G(r)}{1+|\partial_r G(r)|^2} \right) \partial_r.$$
(3.18)

Also the principal curvatures of Σ are given by

$$\kappa_1 = \dots = \kappa_{N-1} = \frac{\partial_r G(r)}{r\sqrt{1 + |\partial_r G(r)|^2}}, \quad \kappa_N = \frac{\partial_{rr} G(r)}{(1 + |\partial_r G(r)|^2)^{\frac{3}{2}}}$$
(3.19)

and

$$|A_{\Sigma}|^2 = \sum_{j=1}^{N} \kappa_j^2 = \frac{(N-1) \left[\partial_r G(r)\right]^2}{r^2 \left(1 + |\partial_r G(r)|^2\right)} + \frac{\left[\partial_{rr} G(r)\right]^2}{\left(1 + |\partial_r G(r)|^2\right)^3}.$$
 (3.20)

Hence, in the coordinates $Y_{\pm}(r,\Theta)$ the Jacobi operator for the surface Σ takes the form

$$\mathcal{J}_{\Sigma} = \Delta_{\Sigma_{\pm}} + |A_{\Sigma_{\pm}}|^{2}
= \frac{\partial_{rr}}{1 + |\partial_{r}G(r)|^{2}} + \left(\frac{N-1}{r} - \frac{\pm \partial_{r}G(r)}{1 + |\partial_{r}G(r)|^{2}}\right) \partial_{r} +
+ \frac{(N-1)[\partial_{r}G(r)]^{2}}{r^{2}(1 + |\partial_{r}G(r)|^{2})} + \frac{[\partial_{rr}G(r)]^{2}}{(1 + |\partial_{r}G(r)|^{2})^{3}}$$
(3.21)

and we recall that we are omitting the explicit dependence of all the quantities in $\varepsilon > 0$.

Next, we consider a dilated version of the geometric objects described above. For $\varepsilon > 0$ small consider the hypersurfaces $\Sigma_{\varepsilon,\pm} := \varepsilon^{-1}\Sigma_{\pm}$ which can be naturally parameterized by

$$Y_{\varepsilon,\pm}(r,\Theta,) = (r\Theta, \pm \varepsilon^{-1}G(\varepsilon r)), \quad r > \varepsilon^{-1}, \quad \Theta \in S^{N-1}.$$

from where we get the associated Fermi coordinates

$$\begin{array}{lcl} X_{\varepsilon,\pm}(r,\Theta) & = & Y_{\varepsilon,\pm}(r,\Theta) \, + \, z \, \nu_{\varepsilon,\pm}(r,\Theta) \\ & = & \frac{1}{\varepsilon} \, Y_{\pm}(\varepsilon r,\Theta) \, + \, z \, \nu_{\pm}(\varepsilon r,\Theta), \quad r > \varepsilon^{-1}, \quad \Theta \in S^{N-1}. \end{array}$$

Observe that in this case the dilated Fermi coordinates $X_{\varepsilon,\pm}(r,\Theta,z)$, provide a local diffeomorphism in a region of the form

$$\mathcal{N}_{\varepsilon,\pm} = \left\{ x = X_{\varepsilon,\pm}(r,\Theta,z) : |z| < \frac{\delta_0}{\varepsilon} \left(1 - \eta_{\varepsilon}(\varepsilon r) \right) + \eta_{\varepsilon}(\varepsilon r) \varepsilon^{-1} F_{\varepsilon}(\varepsilon r) \right\}. \tag{3.22}$$

Next we consider a smooth bounded smooth axially symmetric function $h: \Sigma \to \mathbb{R}$ satisfying the apriori estimate

$$||h||_* := ||D^2 h||_{p,2+\beta} + ||(1+r(y))^{1+\beta} D_{\Sigma} h||_{L^{\infty}(\Sigma)} + ||(1+r(y))^{\beta} h||_{L^{\infty}(\Sigma)} \le C\varepsilon^{\tau}$$
(3.23)

for some p > 1, $\tau > 0$ independent of $\varepsilon > 0$ and where

$$||D^2h||_{p,2+\beta} := \sup_{y \in \Sigma} (1 + r(y)^{2+\beta}) ||D^2h||_{L^p(S_{\Sigma}(y;1))}$$

where $S_{\Sigma}(y;1)$ is the cylinder of width 2 and centered around the sphere passing through $y \in \Sigma$.

We also impose the assumption that h is even over Σ , which is equivalent to saying that, in the coordinates $Y_{\pm}(r,\Theta)$,

$$h(r) := h(Y_{+}(r, \Theta)) = h(Y_{-}(r, \Theta)), \quad r > 1, \quad \theta \in S^{N-1}.$$

Using the translated Fermi coordinates

$$X_{\varepsilon,h}(r,\Theta,t) := X_{\varepsilon}(r,\Theta,t+h(\varepsilon r))$$

and carrying out computations similar to those in section 2 of [1], we find that the Euclidean Laplacian in formula (2.8), when applied to axially symmetric functions, takes the form

$$\Delta_{X_{\varepsilon,h}} = \partial_{tt} + \Delta_{\Sigma_{\varepsilon}} -\varepsilon H_{\Sigma} \partial_{t} - \varepsilon^{2} |A_{\Sigma}|^{2} t \partial_{t} - \varepsilon^{2} \left\{ \Delta_{\Sigma} h + |A_{\Sigma}|^{2} h \right\} \partial_{t} -2\varepsilon \partial_{r} h(\varepsilon r) \partial_{tr} - \varepsilon^{2} [\partial_{r} h(\varepsilon r)]^{2} \partial_{tt} + D_{\varepsilon,h}$$

$$(3.24)$$

where $D_{\varepsilon,h}$ is given by

$$D_{\varepsilon,h} = \varepsilon(t+h)a_{1}(\varepsilon r, \varepsilon(t+h)) \left(\partial_{rr} - 2\varepsilon\partial_{r}h(\varepsilon r)\partial_{rt} - \varepsilon^{2}\partial_{rr}h(\varepsilon r)\partial_{t} + \alpha^{2}[\partial_{r}h(\varepsilon r)]^{2}\partial_{tt}\right) + \varepsilon^{3}(t+h)a_{2}(\varepsilon r, \varepsilon(t+h))\partial_{ij} + \varepsilon^{2}(t+h)b_{1}(\varepsilon r, \alpha(t+h)) \left(\partial_{r} - \varepsilon\partial_{r}h(\varepsilon r)\partial_{t}\right) + \varepsilon^{3}(t+h)^{2}b_{2}(\varepsilon r, \varepsilon(t+h))\partial_{t}$$

$$(3.25)$$

and from (2.3) (2.4) and (2.5), the smooth functions a_1, a_{ij}, b_i and have the asymptotics

$$a_1(r,z) = \mathcal{O}(r^{-2}), \quad a_{ij}(r,z) = \mathcal{O}(r^{-4}), \quad i,j = s_1, \dots, s_{N-1}$$

$$b_1(r,z) = \mathcal{O}(r^{-3}), \quad b_2(r,z) = \mathcal{O}(r^{-6})$$

as $r \to \infty$.

We finish this section with the following remark. For $x = (x', x_{N+1}) \in \mathcal{N}_{\varepsilon, \pm}$ we may write

$$x_{N+1} = X_{\varepsilon,\pm}(r,\Theta,z) \cdot e_{N+1}$$

so that

$$x_{N+1} = \pm \varepsilon^{-1} G(\varepsilon r) + \frac{\pm z}{\sqrt{1 + |\partial_r G(\varepsilon r)|^2}}, \quad x \in \mathcal{N}_{\varepsilon, \pm}$$

which can be rewritten as

$$z = \pm \sqrt{1 + |\partial_r G(\varepsilon r)|^2} \left(x_{N+1} \mp \varepsilon^{-1} G(\varepsilon r) \right)$$
(3.26)

so that, from (3.15), as $\varepsilon r \to \infty$ we have that

$$z \approx \pm \left(x_{N+1} \mp \log \left(\frac{2\sqrt{2} a_0}{\varepsilon^2} \right) \mp 2 \log(\varepsilon r) \pm \log(2(N-2)) \right).$$

4. Jacobi Operator

An important part our developments is to solve the linear equation

$$\mathcal{L}_{\varepsilon}(h) := \varepsilon^{2} \left(\Delta_{\Sigma} h + |A_{\Sigma}|^{2} h \right) + 2\sqrt{2} a_{0} \chi_{\varepsilon} e^{-2\sqrt{2}\varepsilon^{-1} F_{\varepsilon}} h = \varepsilon^{2} g, \quad \text{in } \Sigma$$

$$(4.1)$$

in the class of axially symmetric functions and where the function g satisfies that

$$||g||_{p,\beta} := \sup_{y \in \Sigma} (1 + r(y))^{\beta} ||g||_{L^{p}(S_{\Sigma}(y;1))} < \infty$$

for $\beta > 0$, p > 1 and where $S_{\Sigma}(y; 1)$ is the annulus in Σ centered at $y \in \Sigma$ and width two.

Since F_{ε} is asymptotically logarithmic in $\varepsilon > 0$ and r > 1, and the $N \ge 3$, Σ might be consider as a mild perturbation of the N-catenoid M and therefore the size of $\mathcal{L}_{\varepsilon}^{-1}$ is expected to be uniformly bounded in $\varepsilon > 0$. This is precisely the content of the following Proposition.

Proposition 4.1. For any p and β with p > 1 and $0 < \beta < N-2$ there exists a constant $C = C(p, \beta) > 0$ such that for every $\varepsilon > 0$ and for any axially symmetric function g defined in Σ , that is even respect to the x_{N+1} -axis and with $\|g\|_{p,2+\beta} < \infty$, equation (4.1) has a solution h satisfying the estimate

$$||h||_{2,p,2+\beta} := ||D_{\Sigma}^2 h||_{p,2+\beta} + ||(1+r)^{1+\beta} \nabla_{\Sigma} h||_{L^{\infty}(\Sigma)} + ||(1+r)^{\beta} h||_{L^{\infty}(\Sigma)} \le C ||g||_{p,2+\beta}.$$

This solution is in addition axially symmetric and even respect to the x_{N+1} – axis.

Proof. First recall that $\chi_{\varepsilon}: \mathbb{R}^N \to R$ is given by

$$\chi_{\varepsilon}(r) = \begin{cases} 0, & r > R_{\varepsilon} \\ 1, & r < R_{\varepsilon} \end{cases}$$

Since Σ coincides with the catenoid M for $1 < r < R_{\varepsilon}$ and with the graphs of $\pm F_{\varepsilon}$ for $r > R_{\varepsilon}$, it is natural to look for a solution h of the form

$$h = (1 - \chi_{\varepsilon})h_1 + \chi_{\varepsilon}h_2$$

under the constraints

$$h_1(R_{\varepsilon}) = h_2(R_{\varepsilon}), \quad \partial_r h_1(R_{\varepsilon}) = \partial_r h_2(R_{\varepsilon})$$
 (4.2)

so that h is C^1 over Σ .

We decompose g as well as

$$g = (1 - \chi_{\varepsilon})g + \chi_{\varepsilon}g =: g_1 + g_2$$

and we find that it suffices for h_1 and h_2 to solve the equations

$$\Delta_M h_1 + |A_M|^2 h_1 = g_1, \quad \text{in } M \tag{4.3}$$

$$\varepsilon^{2} \left(\Delta_{\Sigma} h_{2} + |A_{\Sigma}|^{2} h_{2} \right) + 2\sqrt{2} a_{0} e^{-2\sqrt{2}\varepsilon^{-1} F_{\varepsilon}} h_{2} = \varepsilon^{2} g_{2}, \quad \text{in } \Sigma \cap \{r > R_{\varepsilon}\}$$

$$(4.4)$$

with axially symmetric right hand sides g_i satisfying that $||g_i||_{p,2+\beta} < \infty$, i = 1, 2 and under the constraints (4.2).

We begin by studying equation (4.3). Let us remark that the catenoid M can also be parameterized using the local coordinate system

$$(s, \Theta) \in \mathbb{R} \times S^N \mapsto (\phi(s)\Theta, \psi(s)) \in M$$

where the function $\phi(s)$ satisfies the IVP

$$\dot{\phi}^2 + \phi^{4-2N} = \phi^2$$
, $s \in \mathbb{R}$, $\phi(0) = 1$, $\dot{\phi}(0) = 0$

and $\psi(s)$ is given by the rule

$$\dot{\psi} = \phi^{2-N}, \quad \psi(0) = 0.$$

From this we find that ϕ is even and positive and ψ is an odd smooth and increasing diffeomorphism from \mathbb{R} to (-T,T) where we recall that

$$T = \int_{1}^{\infty} \frac{1}{\sqrt{s^{2(N-1)} - 1}} \, ds.$$

Furthermore, there exists a constant a > 0 such that

$$\phi(s) = a e^{|s|} (1 + \mathcal{O}(e^{-2(N-1)|s|})), \quad \text{as } |s| \to \infty.$$
 (4.5)

Through the change of variables $r = \phi(s)$ and setting

$$h_1(r) = \phi^{\frac{2-N}{2}}v(s), \quad g_1(r) = \phi^{-\frac{2+N}{2}}\hat{g}(s)$$
 (4.6)

we find the conjugate form of equation (4.3), namely

$$\partial_{ss}v(s) - \left[\left(\frac{N-2}{2} \right)^2 - \frac{N(3N-2)}{4} \phi^{2-2N} \right] v(s) = \hat{g}(s), \quad s \in \mathbb{R}.$$
 (4.7)

Since we are in the axially symmetric even setting, we consider equation (4.7) with the initial condition $\partial_s v(0) = 0$.

We know that the kernel associated to (4.7) contains at least the jacobi fields coming from dilations and translations along the coordinate axis, i.e. when looked through local coordinates, the functions

$$y \cdot \nu(y), \quad \nu(y) \cdot e_j, \quad j = 1, \dots, N+1, \quad y \in M$$

are elements of the kernel of equation (4.7). We remark also that jacobi field coming from rotations around the axis of symmetry is zero.

We readily check that

$$z_1(y) = \nu(y) \cdot e_{N+1}, \quad z_2(y) = y \cdot \nu(y), \quad y \in M$$

are the only axially symmetric ones and in terms of the functions $\phi(s)$ and $\psi(s)$ we have that

$$z_1(s) = \phi^{\frac{N-4}{2}} \partial_s \phi, \quad z_2(s) = \phi^{\frac{N-4}{2}} (\partial_s \phi \psi - T \partial_s \psi \phi)$$

from where it directly follows that z_1 is odd and z_2 is even. Also from the asymptotics in (4.5) it follows that

$$z_1(s) \sim e^{\frac{N-2}{2}s}, \quad z_2(s) \sim e^{-\frac{N-2}{2}s}, \quad s \to \infty$$

with wronskian equals 1.

Hence, using the variation of parameter formula we set

$$v(\mathbf{s}) = -z_1(\mathbf{s}) \int_0^{\mathbf{s}} z_2(\varsigma) \hat{g}(\varsigma) d\varsigma + z_2(\mathbf{s}) \int_0^{\mathbf{s}} z_1(\varsigma) \hat{g}(\varsigma) d\varsigma. \tag{4.8}$$

Next, we estimate the size of v. For any $s_0 > 0$, $r_0 = \phi(s_0) > 1$, and for any $j_0 \in \mathbb{N}$ such that $j_0 \leq r_0 \leq j_0 + 1$, we estimate

$$\left| \int_{0}^{s_{0}} z_{2}(\varsigma) \hat{g}(\varsigma) d\varsigma \right| \leq C \int_{0}^{s_{0}} e^{-\frac{N-2}{2}\varsigma} e^{\frac{N+2}{2}\varsigma} |g_{1}(\phi(\varsigma))| d\varsigma$$

$$\leq C \int_{1}^{r_{0}} \xi^{N-1} \xi^{-(N-2)} |g_{1}(\xi)| d\xi$$

$$\leq C \sum_{j=1}^{j_{0}} \int_{j}^{j+1} \xi^{N-1} \xi^{-(N-2)} |g_{1}(\xi)| d\xi \leq C \sum_{j=1}^{j_{0}} j^{-(N-2)} \int_{j}^{j+1} \xi^{N-1} |g_{1}(\xi)| d\xi$$

$$\leq C \sum_{j=1}^{j_{0}} j^{-(N-2)-2-\beta+(N-1)\frac{1}{p'}} j^{2+\beta} ||g_{1}||_{L^{p}(S_{\Sigma}(Y(j,\theta);1))}$$

$$\leq C ||g_{1}||_{p,2+\beta} \sum_{j=1}^{j_{0}} j^{-(N-1)\frac{1}{p}-1-\beta} \leq C ||g_{1}||_{p,2+\beta} (1+r_{0})^{-(N-1)\frac{1}{p}-\beta}$$

$$\leq C ||g_{1}||_{p,2+\beta} e^{[-(N-1)\frac{1}{p}-\beta]s_{0}}.$$

Proceeding in the same fashion we estimate the second integral in (4.8) to obtain that

$$\begin{split} \left| \int_{0}^{s_{0}} z_{1}(\varsigma) \hat{g}(\varsigma) d\varsigma \right| & \leq C \int_{0}^{s_{0}} e^{-\frac{N-2}{2}\varsigma} e^{\frac{N+2}{2}\varsigma} |g_{1}(\phi(\varsigma))| d\varsigma \leq C \int_{0}^{s_{0}} e^{(N-1)\varsigma} |g_{1}(\phi(\varsigma))| e^{\varsigma} d\varsigma \\ & \leq C \int_{1}^{r_{0}} \xi^{N-1} |g_{1}(\xi)| d\xi \leq C \|g_{1}\|_{p,2+\beta} \sum_{j=1}^{j_{0}} j^{(N-1)\frac{1}{p'}-2-\beta} \\ & \leq C \|g_{1}\|_{p,2+\beta} r_{0}^{-(N-1)\frac{1}{p}-\beta+N-2} \approx C \|g_{1}\|_{p,2+\beta} e^{[-(N-1)\frac{1}{p}-\beta+N-2]s_{0}}. \end{split}$$

From (4.6) and the previous estimates, it is direct to check that

$$\|(1+r(y)^{1+\beta})D_M h_1\|_{L^{\infty}(M)} + \|(1+r(y)^{\beta})h_1\|_{L^{\infty}(M)} \le C\|g_1\|_{p,2+\beta}$$

Using elliptic estimates and directly from equation (4.3) we obtain for every p > 1 and $\beta > 0$ that

$$||h_1||_{2,p,2+\beta} \le C||g_1||_{2+\beta}. \tag{4.9}$$

Next, we study equation (4.4)

$$\varepsilon^2 \left(\Delta_{\Sigma} h_2 + |A_{\Sigma}|^2 h_2 \right) + 2\sqrt{2}a_0 e^{-2\sqrt{2}\varepsilon^{-1}F_{\varepsilon}} h_2 = g_2, \quad r > R_{\varepsilon}$$

with the initial conditions

$$h_2(R_{\varepsilon}) = h_1(R_{\varepsilon}), \quad \partial_r h_2(R_{\varepsilon}) = \partial_r h_1(R_{\varepsilon})$$

which can be estimated from (4.9) to obtain that

$$|R_{\varepsilon}^{1+\beta} \, \partial_r h_1(R_{\varepsilon})| + |R_{\varepsilon}^{\beta} \, h_1(R_{\varepsilon})| \le C \|g_1\|_{p,2+\beta}.$$

Expression (3.15) implies that for $r > R_{\varepsilon}$

$$2\sqrt{2}a_0e^{-2\sqrt{2}\varepsilon^{-1}F_{\varepsilon}} = \frac{2(N-2)\varepsilon^2}{r^2}e^{\bar{\omega}_{\varepsilon}}$$
$$= \frac{2(N-2)\varepsilon^2}{r^2} + \varepsilon^2\mathcal{O}(r^{-2}\bar{\omega}_{\varepsilon}).$$

Observe also that (3.15) together with (3.20) and (3.21) imply that

$$\Delta_{\Sigma} = \Delta_{\mathbb{R}^N} + \varepsilon^2 \mathcal{O}(r^{-2}) \partial_{rr} + \varepsilon \mathcal{O}(r^{-1}) \partial_r$$

and

$$c \varepsilon^2 (1+r)^{-4} \le |A_{\Sigma}|^2 \le C \varepsilon^2 (1+r)^{-4}, \quad r > R_{\varepsilon}.$$

Therefore, the model linear equation associated to (4.4) is

$$\Delta_{\mathbb{R}^N} \hat{h} + \frac{2(N-2)}{r^2} (1 + p(r)) \hat{h} = \hat{g}, \quad r > R_{\varepsilon}$$

where

$$p(r) = \frac{2(N-2)}{r^2} \left(e^{-\bar{\omega}_{\varepsilon}} - 1 \right) \approx 2(N-2)r^{-2}\bar{\omega}_{\varepsilon}, \text{ as } r \to \infty$$

and whose linear differential operator resembles a Hardy operator at infinity.

Abusing the notation and making the scaling

$$\hat{h}(r) \sim \hat{h}\left(\frac{r}{R_{\varepsilon}}\right)$$

we find that it suffices to solve the problem

$$\Delta_{\mathbb{R}^N} \hat{h} + \frac{2(N-2)}{r^2} (1 + p(r)) \hat{h} = R_{\varepsilon}^2 \hat{g}(r), \quad r > 1$$
(4.10)

$$\hat{h}(1) = h_1(R_{\varepsilon}), \quad \partial_r \hat{h}(1) = R_{\varepsilon} \partial_r h_1(R_{\varepsilon})$$

$$(4.11)$$

where

$$|\partial_r \hat{h}(1)| \,+\, |\hat{h}(1)| \leq C\, R_\varepsilon^{-\beta}\, \|g_1\|_{p,2+\beta}.$$

From the classic theory in asymptotic integration of second order linear ODE, described for instance in chapter 11 in [22] or following the same lines of the proof of lemma 4.1 in [1], we find two smooth linearly independent elements of the kernel for (4.10) $z_{\pm}(r)$, such that in the radial variable they have the asymptotics at infinity

$$z_{+}(r) \approx r^{-\frac{N-2}{2}} \begin{cases} \cos\left(\sqrt{(N-2)(10-N)}\log(r)\right), & 3 \leq N \leq 9 \\ \frac{1, \quad N = 10}{r^{+\frac{1}{2}}\sqrt{(N-2)(N-10)}}, \quad N \geq 11 \end{cases}$$

$$z_{-}(r) \approx r^{-\frac{N-2}{2}} \begin{cases} \sin\left(\sqrt{(N-2)(10-N)}\log(r)\right), & 3 \leq N \leq 9 \\ \frac{\log(r), \quad N = 10}{r^{-\frac{1}{2}}\sqrt{(N-2)(N-10)}}, \quad N \geq 11 \end{cases}$$

and these relations can be differentiated. Also we can compute the wronskian $W(r) = W(z_+, z_-)$ to find the asymptotics

$$r^{N-1}W(r) = \begin{cases} -\sqrt{(N-2)(10-N)}, & 3 \le N \le 9\\ -1, & N = 10\\ -\sqrt{(N-2)(N-10)}, & N \ge 11. \end{cases}$$
 (4.12)

Then we set as solution the function

$$\hat{h}(r) = A z_{+}(r) + B z_{-}(r) + \hat{z}(r) \tag{4.13}$$

where $\hat{z}(r)$ is the particular solution

$$\hat{z}(r) = -z_{+}(r) \int_{1}^{r} \varsigma^{N-1} z_{-}(\varsigma) \hat{g}(\varsigma) d\varsigma + z_{-}(r) \int_{1}^{r} \varsigma^{N-1} z_{+}(\varsigma) \hat{g}(\varsigma) d\varsigma.$$

Since

$$\hat{h}(1) = Az_{+}(1) + Bz_{-}(1)$$
$$\partial_{r}\hat{h}(1) = A \,\partial_{r}z_{+}(1) + B \,\partial_{r}z_{-}(1)$$

we find that

$$\begin{pmatrix} A \\ B \end{pmatrix} = W(1)^{-1} \begin{pmatrix} z_{+}(1) & z_{-}(1) \\ \partial_{r}z_{+}(1) & \partial_{r}z_{-}(1) \end{pmatrix} \begin{pmatrix} \hat{h}(1) \\ \partial_{r}\hat{h}(1) \end{pmatrix}$$

and using (4.11) and (4.12) we find that

$$|A| + |B| \le C R_{\varepsilon}^{-\beta} \|g_1\|_{p,2+\beta}.$$

In order to estimate the function $\hat{z}(r)$ we proceed in the exact same fashion as above to find that

$$\|\hat{z}\|_{2,p,2+\beta} \le C R_{\varepsilon}^{-\beta} \|g_2\|_{p,2+\beta}$$

Rescaling back we obtain that

$$h_2(r) = Az_+\left(\frac{r}{R_{\varepsilon}}\right) + Bz_-\left(\frac{r}{R_{\varepsilon}}\right) + \hat{z}\left(\frac{r}{R_{\varepsilon}}\right), \quad r > R_{\varepsilon}$$

from where we find that

$$||h_2||_{2,p,2+\beta} \le C||g_2||_{p,2+\beta}$$

provided that $0 < \beta < \frac{N-2}{2}$ which is a slower decay than the one of the functions $z_{\pm}(r)$.

To finish the proof, let us denote by $\mathcal{L}(g_2) := h_2$ the resolvent operator for the model linear equation (4.10). We apply a fixed point argument to the expression

$$\mathcal{L}\left[g_2 - |A_{\Sigma}|^2 h_2 + (\Delta_{\Sigma} - \Delta_{\mathbb{R}^N}) h_2\right] = h_2$$

in the topologies described above, so that for $\varepsilon > 0$ small enough we find a unique solution to the equation satisfying the same estimate and this completes the proof of the Proposition.

5. Approximation and preliminary discussion

Now we are in position to describe the approximation we want to consider. Denote by w(t) the heteroclinic solution to the one dimensional Allen-Cahn equation

$$w'' + w(1 - w^2) = 0$$
, in \mathbb{R} , $w(\pm \infty) = \pm 1$, $w'(t) > 0$ (5.1)

which is given explicitly by

$$w(t) = \tanh\left(\frac{t}{\sqrt{2}}\right), \quad t \in \mathbb{R}$$

and has the following asymptotic behavior

$$w(t) = \begin{cases} 1 - 2e^{-\sqrt{2}t} + \mathcal{O}(e^{-2\sqrt{2}t}), & t > 0\\ -1 + 2e^{\sqrt{2}t} + \mathcal{O}(e^{2\sqrt{2}t}), & t < 0 \end{cases}$$
 (5.2)

and relation (5.2) can be differentiated so that

$$w'(t) = 2\sqrt{2}e^{-\sqrt{2}|t|} + \mathcal{O}(e^{-2\sqrt{2}|t|}), \quad t \in \mathbb{R}.$$

$$w''(t) = -4e^{-\sqrt{2}|t|} + \mathcal{O}(e^{-2\sqrt{2}|t|}), \quad t \in \mathbb{R}.$$

From (3.22) we consider the set $\mathcal{N}_{\varepsilon} = \mathcal{N}_{\varepsilon,+} \cup \mathcal{N}_{\varepsilon,-}$, the cylinder

$$C_{R_{\varepsilon}} := \left\{ y \in \varepsilon^{-1} \Sigma : r(\varepsilon y) < R_{\varepsilon} \right\}.$$

and the cut-off function $\eta_{\varepsilon}(r) := \eta(\varepsilon r - R_{\varepsilon})$ with

$$\eta(s) = \begin{cases} 0, & s < -1 \\ 1, & s > 1. \end{cases}$$

Recall also that in sections 2 and 3 we introduced a smooth bounded parameter function h satisfying the apriori estimate (3.23).

We define our approximation to (1.1) in the region $\mathcal{N}_{\varepsilon}$ as the function

$$\bar{u}(x) = (1 - \eta_{\varepsilon})u_0 + \eta_{\varepsilon}u_1 \tag{5.3}$$

where

$$u_0(x) = w(z - h(\varepsilon r)), \quad x \in \mathcal{N}_{\varepsilon} \cap C_{R_{\varepsilon}}$$
 (5.4)

and taking into account (3.26)

$$u_1(x) = w(\sqrt{1 + |\partial_r F_{\varepsilon}(\varepsilon r)|^2} (x_{N+1} - \varepsilon^{-1} F_{\varepsilon}(\varepsilon r)) - h(\varepsilon r))$$

$$-w(\sqrt{1+|\partial_r F_{\varepsilon}(\varepsilon r)|^2}\left(x_{N+1}+\varepsilon^{-1}F_{\varepsilon}(\varepsilon r)\right)+h(\varepsilon r))-1, \quad x \in \mathcal{N}_{\varepsilon}-C_{R_{\varepsilon}}. \quad (5.5)$$

Observe that in $\mathcal{N}_{\varepsilon} \cap C_{R_{\varepsilon}}$ the function u_0 coincides with the heteroclinic solution while, from (3.26)

$$u_1(x) \approx w(x_{N+1} - \varepsilon^{-1} F_{\varepsilon}(\varepsilon r) - h(\varepsilon r)) - w(z + \varepsilon^{-1} F_{\varepsilon}(\varepsilon r) + h(\varepsilon r)) - 1$$

for $x = y + z\nu_{\varepsilon}(y) \in \mathcal{N}_{\varepsilon}$ with $r(\varepsilon y) > R_{\varepsilon}$.

Since $\mathbb{R}^{N+1} - \Sigma = S_+ \cup S_-$, we also have the associated dilated version $\mathbb{R}^{N+1} - \Sigma_{\varepsilon} = S_{\varepsilon,+} \cup S_{\varepsilon,-}$ where $S_{\varepsilon,\pm} := \varepsilon^{-1} S_{\pm}$ respectively. Consider the function \mathbb{H} given by

$$\mathbb{H}(x) = \begin{cases} 1, & \text{in } S_{\varepsilon,+} \\ -1, & \text{in } S_{\varepsilon,-}. \end{cases}$$

As global approximation we take the interpolation between the functions \bar{u} and \mathbb{H} . To be precise we use a cut-off function ρ_{ε} described by

$$\rho_{\varepsilon}(x) = \eta \left(|z| - \frac{\delta_0}{\varepsilon} (1 - \chi_{\varepsilon}) + \chi_{\varepsilon} \varepsilon^{-1} F_{\varepsilon}(\varepsilon r) + 1 \right), \quad x = X_{\varepsilon}(r, \Theta, z) \in \mathcal{N}_{\varepsilon}$$

to write

$$\mathbf{w}_{\varepsilon}(x) = \rho_{\varepsilon}\bar{u}(x) + (1 - \rho_{\varepsilon})\mathbb{H}(x), \quad \text{in } \mathbb{R}^{N+1}.$$
 (5.6)

Our next and crucial step is to compute the error of this approximation. Let us write

$$S(\mathbf{w}_{\varepsilon}) := \Delta \mathbf{w}_{\varepsilon} + F(\mathbf{w}_{\varepsilon})$$

where

$$F(\mathbf{w}_{\varepsilon}) := \mathbf{w}_{\varepsilon}(1 - \mathbf{w}_{\varepsilon}^2).$$

Observe that $S(\mathbf{w}_{\varepsilon})$ splits into

$$S(\mathbf{w}_{\varepsilon}) = \underbrace{\rho_{\varepsilon}(x)S(\bar{u})}_{I} + \underbrace{2\rho_{\varepsilon} \cdot \nabla(\bar{u} - \mathbb{H}) + (\bar{u} - \mathbb{H})\Delta\rho_{\varepsilon}}_{II} + \underbrace{F(\mathbf{w}_{\varepsilon}) - \rho_{\varepsilon}F(\bar{u})}_{III}$$
(5.7)

so that we compute the sizes of each of the terms involved in expression (5.7).

First we compute I. Using the characteristic function χ_{ε} as in (3.17)

$$S(\bar{u}) = (1 - \chi_{\varepsilon}(\varepsilon r))S(\bar{u}) + \chi_{\varepsilon}(\varepsilon r)S(\bar{u}). \tag{5.8}$$

Setting $z = t + h(\varepsilon r)$, we find from expression (3.24) that in the region $\mathcal{N}_{\varepsilon,h} \cap C_{R_{\varepsilon}}$, $\bar{u}(x) = w(t)$. So that $S(\bar{u}(x)) = -\varepsilon H_{\Sigma}(\varepsilon r)w'(t) - \varepsilon^2 |A_{\Sigma}(\varepsilon r)|^2 t w'(t)$

$$-\varepsilon^{2} \left\{ \Delta_{\Sigma} h + |A_{\Sigma}|^{2} h \right\} w'(t) - \varepsilon^{2} \left[\partial_{r} h(\varepsilon r) \right]^{2} w''(t)$$

$$-\varepsilon^{3}(t+h)a_{1}(\varepsilon r,\varepsilon(t+h))\left(\partial_{rr}h(\varepsilon r)w'(t)+\left[\partial_{r}h(\varepsilon r)\right]^{2}w''(t)\right)$$

$$-\varepsilon^{3}(t+h)b_{1}(\varepsilon r,\varepsilon(t+h))\partial_{r}h(\varepsilon r)w'(t)+\varepsilon^{3}(t+h)^{2}b_{N+1}(\varepsilon r,\varepsilon(t+h))w'(t).$$
(5.9)

The minimality of M implies that $H_{\Sigma} = H_M = 0$ in the cylinder $C_{R_{\varepsilon}}$ and we conclude that

$$|S(\bar{u}(x)) + \mathcal{J}_{\Sigma}(h)(\varepsilon r)w'(t)| \le C\varepsilon^2(1+\varepsilon r)^{-2}e^{-\sigma|t|}$$

for any $0 < \sigma < \sqrt{2}$.

As for the second term in (5.8), setting

$$t = \sqrt{1 + |\partial_r G(\varepsilon r)|^2} \left(x_{N+1} - \varepsilon^{-1} \varepsilon^{-1} G_{\varepsilon}(\varepsilon r) - h(\varepsilon r) \right)$$
$$= z - h(\varepsilon r)$$

in the set $\Sigma_{\varepsilon,+}$ for $r > R_{\varepsilon}$, we observe that

$$\bar{u}(x) = w(t) - w(t + 2\varepsilon^{-1}F_{\varepsilon}(\varepsilon r) + 2h(\varepsilon r)) - 1$$

and consequently

$$F(\bar{u}) = F(w(t)) - F(w(t + 2\varepsilon^{-1}F_{\varepsilon}(\varepsilon r) + 2h(\varepsilon r))) - 1 + 1$$
$$- (F'(w(t)) - F'(-1)) [w(t + 2\varepsilon^{-1}F_{\varepsilon} + 2h(\varepsilon r)) + 1]$$
$$+ \frac{1}{2} (F''(w(t)) + F'(-1)) [w(t + 2\varepsilon^{-1}F_{\varepsilon} + 2h(\varepsilon r)) + 1]^{2} + \mathcal{O}\left([w(t + 2F_{\varepsilon} + 2h(\varepsilon r)) + 1]^{3}\right).$$

From the asymptotics in (5.2) and since h is even in Σ , we obtain that

$$F(\bar{u}) = F(w(t)) - F(w(t + 2\varepsilon_{\varepsilon}^{-1} + 2h)) - 6(1 - w^{2}(t))e^{\sqrt{2}t}e^{-2\sqrt{2}(\varepsilon^{-1}F_{\varepsilon} + h)} + 6\left[(1 - w^{2}(t)) + 2(1 - w(t))\right]e^{2\sqrt{2}t}e^{-2\sqrt{2}(\varepsilon^{-1}F_{\varepsilon} + h)} + \mathcal{O}\left(e^{-3\sqrt{2}|t + 2\varepsilon^{-1}F_{\varepsilon} + 2h|}\right). \quad (5.10)$$

Let us now consider the decomposition

$$6(1 - w^{2}(t))e^{-\sqrt{2}t} = a_{0} w'(t) + g_{0}(t), \quad \int_{\mathbb{R}} g_{0}(t)w'(t)dt = 0.$$
 (5.11)

Using the function g_0 and (5.10) we find that in $\Sigma_{\varepsilon,+}$ and for $r > R_{\varepsilon}$

$$S(\bar{u}) = -\left(\varepsilon H_{\Sigma}(\varepsilon r) - a_{0} e^{-2\sqrt{2}\varepsilon^{-1}F_{\varepsilon}}\right) w'(t) - \varepsilon^{2} |A_{\Sigma}|^{2} t \, w'(t) + g_{0}(t)e^{-2\sqrt{2}\varepsilon^{-1}F_{\varepsilon}}$$

$$-\varepsilon^{2} \left(\Delta_{\Sigma} h + |A_{\Sigma}|^{2} h\right) w'(t) - 6\left(1 - w^{2}(t)\right) e^{\sqrt{2}t} e^{-2\sqrt{2}\varepsilon^{-1}F_{\varepsilon}} \left[e^{-2\sqrt{2}h} - 1\right]$$

$$+\varepsilon^{2} |\nabla_{\Sigma} h|^{2} w''(t) + \varepsilon^{2} \left(\Delta_{\Sigma} h + |A_{\Sigma}|^{2} h\right) w'(t + 2\varepsilon^{-1}F_{\varepsilon} + 2h) + \varepsilon^{2} |\nabla_{\Sigma} h|^{2} w''(t - 2f_{\alpha})$$

$$+ \left[6(1 - w^{2}(t)) + 12(1 - w(t))\right] e^{2\sqrt{2}t} e^{-4\sqrt{2}(\varepsilon^{-1}F_{\varepsilon} + h)} + \widetilde{R}$$
(5.12)

and

$$|D_p\widetilde{\mathbf{R}}(\varepsilon r, t, p, q)| + |D_q\widetilde{\mathbf{R}}(\varepsilon r, t, p, q)| + |\widetilde{\mathbf{R}}(\varepsilon r', t, p, q)| \le C\varepsilon^{2+\tau} (1 + |\varepsilon r|)^{-4} e^{-\sigma|t|}$$

$$(5.13)$$

for any $0 < \sigma < \sqrt{2}$ and any $0 < \tau < 1$. Similar computations are obtained in the set $\Sigma_{\varepsilon,-}$ for $r > R_{\varepsilon}$.

The asymptotics for F_{ε} in (3.15) imply that for $r > R_{\varepsilon}$

$$H_{\Sigma} = \nabla \cdot \left(\frac{\nabla F_{\varepsilon}}{\sqrt{1 + |\nabla F_{\varepsilon}|^2}} \right) = \Delta_{\mathbb{R}^N} F_{\varepsilon} + \varepsilon^3 \mathcal{O}((1+r)^{-4}).$$

Using the fact that the function F_{ε} is an exact solution of equation (3.3) we can write

$$E_0(\varepsilon r) := \varepsilon H_{\Sigma}(\varepsilon r) - a_0 e^{-2\sqrt{2}\varepsilon^{-1}F_{\varepsilon}} = \varepsilon^4 \mathcal{O}((1+\varepsilon r)^{-4})$$
(5.14)

and from (5.14) we observe that for $\varepsilon r > R_{\varepsilon}$

$$S(\bar{u}) = -\varepsilon^{2} \left(\Delta_{\Sigma} h + |A_{\Sigma}|^{2} h \right) w'(t) - 6 \left(1 - w^{2}(t) \right) e^{\sqrt{2}t} e^{-2\sqrt{2}\varepsilon^{-1}F_{\varepsilon}} \left[e^{-2\sqrt{2}h} - 1 \right]$$

$$-E_0(\varepsilon r)w'(t) - \varepsilon^2 |A_{\Sigma}|^2 t \, w'(t) + g_0(t)e^{-2\sqrt{2}\varepsilon^{-1}F_{\varepsilon}} + \varepsilon^2 |\nabla_{\Sigma} h|^2 \, w''(t)$$

$$+ \varepsilon^{2} \left(\Delta_{\Sigma} h + |A_{\Sigma}|^{2} h \right) w'(t + 2\varepsilon^{-1} F_{\varepsilon} + 2h) + \varepsilon^{2} |\nabla_{\Sigma} h|^{2} w''(t - 2f_{\alpha}) + \widetilde{R}$$

$$(5.15)$$

where \widetilde{R} is as described in (5.13).

From this we find that for $\varepsilon r > R_{\varepsilon}$

$$|S(\bar{u})\chi_{\varepsilon}| \le C\varepsilon^{2-\tau}(1+\varepsilon r)^{-2+\tau}e^{-\sigma|t|}$$

From (5.9) and (5.15) that for any $0 < \sigma < \sqrt{2}$ and any $\tau = \frac{\sigma}{\sqrt{2}}$

$$||S(\bar{u})||_{p,2-\tau,\sigma} \le C\varepsilon^{2-\tau}.$$

Since the cut-off function ρ_{ε} is supported in a region of the form

$$\frac{\delta_0}{\varepsilon}(1-\chi_{\varepsilon}) + \chi_{\varepsilon}\varepsilon^{-1}F_{\varepsilon}(\varepsilon r) - 1 \le |z| \le \frac{\delta_0}{\varepsilon}(1-\chi_{\varepsilon}) + \chi_{\varepsilon}\varepsilon^{-1}F_{\varepsilon}(\varepsilon r) + 1$$

the contribution of the term II in (5.7) is

$$|2\nabla \rho_{\varepsilon} \cdot \nabla \bar{u}(x) + (\bar{u}(x) - \mathbb{H}(x))\Delta \rho_{\varepsilon}| \le C e^{\sqrt{2}|t + 2\chi_{\varepsilon}\varepsilon^{-1}F_{\varepsilon} + 2h|}.$$

As for the term II in (5.7), we can easily check that

$$III = 3(\bar{u} - \mathbb{H})^2 \rho_{\varepsilon} (1 - \rho_{\varepsilon}) + 3(\bar{u} - \mathbb{H})^3 \rho_{\varepsilon} (1 - \rho_{\varepsilon}^3)$$

so that

$$|II| + |III| \le Ce^{-\sqrt{2}|t|} \le Ce^{-\sigma|t|} \le C\varepsilon^{2-\frac{\sigma}{\sqrt{2}}} (1+\varepsilon r)^{-2+\frac{\sigma}{\sqrt{2}}} e^{-\sigma|t|}.$$

Putting together the estimates for I, II, III, we find that

$$||S(\mathbf{w}_{\varepsilon})||_{p,2-\tau,\sigma} \le C\varepsilon^{2-\tau} \tag{5.16}$$

for any $\sigma \in (0, \sqrt{2})$ and any $\tau = \frac{\sigma}{\sqrt{2}}$.

We remark also that in the upper part of the set $\mathcal{N}_{\varepsilon,h}$

$$S(\bar{u}) = -\varepsilon^{2} \{ \Delta_{\Sigma} h + |A_{\Sigma}|^{2} h \} w'(t) - 6 (1 - w^{2}(t)) e^{\sqrt{2}t} \chi_{\varepsilon} 2\sqrt{2} e^{-2\sqrt{2}\varepsilon^{-1}F_{\varepsilon}} h \}$$

$$-\chi_{\varepsilon} E_0(\varepsilon r) w'(t) + g_0(t) e^{-2\sqrt{2}\varepsilon^{-1}F_{\varepsilon}} - \varepsilon^2 |A_{\Sigma}|^2 t \, w'(t) + \varepsilon^2 |\nabla_{\Sigma} h|^2 \, w''(t)$$

$$-6(1 - w^{2}(t))e^{\sqrt{2}t}\chi_{\varepsilon}(\varepsilon r) e^{-2\sqrt{2}\varepsilon^{-1}F_{\varepsilon}} \left[e^{-2\sqrt{2}h} - 1 + 2\sqrt{2}h \right] + R.$$

and where

$$|D_p R(\varepsilon r, t, p, q)| + |D_q R(\varepsilon r, t, p, q)| + |R(\varepsilon r', t, p, q)| \le C\varepsilon^{2-\tau} (1 + |\varepsilon r|)^{-2+\tau} e^{-\sigma|t|}$$
(5.17)

for some $0 < \sigma < \sqrt{2}$ and some $\tau = \frac{\sigma}{\sqrt{2}}$.

6. Lyapunov reduction Scheme

We begin this section by setting up the functional analytic spaces we need to carry out the proof of Theorem 1.

Let us first consider the function $r: \mathbb{R}^{N+1} \to [0, \infty)$ given by

$$r(x', x_{N+1}) := |y'|, \quad x = (x', x_{N+1}) \in \mathbb{R}^{N+1}$$

Let us define for $\varepsilon > 0$, $\mu > 0$ and f(x), defined in \mathbb{R}^{N+1} , the norm

$$||f||_{p,\mu,\sim} := \sup_{x \in \mathbb{R}^{N+1}} (1 + r(\varepsilon x))^{\mu} ||f||_{L^p(B_1(x))}, \quad p > 1.$$
(6.1)

We also consider $0 < \sigma < \sqrt{2}$, $\mu > 0$, $\varepsilon > 0$ and functions g = g(y,t) and $\phi = \phi(y,t)$, defined for every $(y,t) \in \Sigma_{\varepsilon} \times \mathbb{R}$. Let us set the norms

$$||g||_{p,\mu,\sigma} := \sup_{(y,t)\in\Sigma_{\varepsilon}\times\mathbb{R}} (1 + r(\varepsilon y))^{\mu} e^{\sigma|t|} ||g||_{L^{p}(B_{1}(y,t))}$$
(6.2)

$$\|\phi\|_{\infty,\mu,\sigma} := \|(1 + r(\varepsilon y)^{\mu})e^{\sigma|t|}\phi\|_{L^{\infty}(\Sigma_{\varepsilon} \times \mathbb{R})}$$

$$(6.3)$$

$$\|\phi\|_{2,p,\mu,\sigma} := \|D^2 \phi\|_{p,\mu,\sigma} + \|D\phi\|_{\infty,\mu,\sigma} + \|\phi\|_{\infty,\mu,\sigma}. \tag{6.4}$$

Finally, for functions h and \tilde{g} defined in Σ , we recall the norms

$$\|\widetilde{g}\|_{p,\beta} := \sup_{y \in \Sigma} (1 + r(y))^{\beta} \|\widetilde{g}\|_{L^{p}(S_{\Sigma}(y;1))}$$
(6.5)

$$||h||_{2,p,\beta} := ||D^2 h||_{p,\beta} + ||(1+r(y)) D h||_{L^{\infty}(\Sigma)} + ||h||_{L^{\infty}(\Sigma)}.$$
(6.6)

Now, in order to prove Theorem 1, let us look for a solution to equation (1.1) of the form

$$u_{\varepsilon}(x) = w_{\varepsilon}(x) + \varphi(x)$$

where $w_{\varepsilon}(x)$ is the global approximation defined in (5.6) and φ is going to be chosen small in an appropriate topology.

Denoting $F(u) = u(1-u^2)$, we observe that for $u_{\varepsilon}(x)$ to be a solution to (1.1), the function φ must solve the equation

$$\Delta \varphi + F'(\mathbf{w}_{\varepsilon})\varphi + S(\mathbf{w}_{\varepsilon}) + N(\varphi) = 0, \text{ in } \mathbb{R}^{N+1}$$

or equivalently

$$\Delta \varphi + F'(\mathbf{w}_{\varepsilon})\varphi = -S(\mathbf{w}_{\varepsilon}) - N(\varphi) \tag{6.7}$$

where

$$N(\varphi) = F(\mathbf{w}_{\varepsilon} + \varphi) - F(\mathbf{w}_{\varepsilon}) - F'(\mathbf{w}_{\varepsilon}).$$

6.1 Gluing procedure: The strategy here consists in transforming (6.7) into a system of two PDEs. One of these equations will take account of the geometry of Σ and w_{ε} near Σ_{ε} , while the other will handle the situation far away from Σ_{ε} , where w_{ε} is almost constant.

In order to proceed, we consider again the non-negative cut-off function $\zeta \in C^{\infty}(\mathbb{R})$ such that

$$\zeta(s) = \begin{cases} 1, & s < -1 \\ 0, & s > 1 \end{cases}$$

and define for $n \in \mathbb{N}$, the cut off function for $x = X_{\varepsilon,\pm,h}(y,t) \in \mathcal{N}_{\varepsilon,\pm,h}$

$$\zeta_n(x) = \zeta \left(|t + h| - \frac{\delta_0}{\varepsilon} \left(1 - \eta_{\varepsilon} \right) - \eta_{\varepsilon} \varepsilon^{-1} F_{\varepsilon}(\varepsilon r) + n \right)$$
(6.8)

for $x = X_{\varepsilon,\pm,h}(r,\Theta,t) \in \mathcal{N}_{\varepsilon,\pm,h}$.

Therefore, we look for a solution $\varphi(x)$ with the particular form

$$\varphi(x) = \zeta_3(x) \phi(y,t) + \psi(x)$$

where the function $\phi(y,t)$ is defined in $\Sigma_{\varepsilon} \times \mathbb{R}$ and the function $\psi(x)$ is defined in the whole \mathbb{R}^{N+1} .

From equation (6.7) and noticing that $\zeta_2 \cdot \zeta_3 = \zeta_3$, we find that

$$\zeta_{3} \left[\Delta_{\mathcal{N}_{\varepsilon,h}} \phi + F'(\zeta_{2} \mathbf{w}_{\varepsilon}) \phi + \zeta_{2} S(\mathbf{w}_{\varepsilon}) + \zeta_{2} N(\phi + \psi) + \zeta_{2} (F'(\mathbf{w}_{\varepsilon}) + 2) \psi \right]$$

$$+ \Delta \psi - \left[2 - (1 - \zeta_{3}) (F'(\mathbf{w}_{\varepsilon}) + 2) \right] \psi + (1 - \zeta_{3}) S(\mathbf{w}_{\varepsilon})$$

$$+ \nabla \zeta_{3} \cdot \nabla_{\mathcal{N}_{\varepsilon,h}} \phi + \phi \Delta \zeta_{3} + (1 - \zeta_{3}) N[\psi + \zeta_{2} \phi] = 0.$$

Hence, to solve (6.7), it suffices to solve the system of PDEs

$$\Delta \psi - \left[2 - (1 - \zeta_2)(F'(\mathbf{w}_{\varepsilon}) + 2)\right] \psi = -(1 - \zeta_2) S(\mathbf{w}_{\varepsilon}) - 2 \nabla \zeta_2 \cdot \nabla_{\mathcal{N}_{\varepsilon,h}} \phi - \phi \Delta \zeta_2 - (1 - \zeta_3) N \left[\zeta_2 \phi + \psi\right], \quad \text{in } \mathbb{R}^{N+1} \quad (6.9)$$

$$\Delta_{\mathcal{N}_{\varepsilon,h}}\phi + F'(\zeta_2 \mathbf{w}_{\varepsilon})\phi = -\zeta_2 S(\mathbf{w}_{\varepsilon}) - \zeta_2 N(\phi + \psi) - \zeta_2 (F'(\mathbf{w}_{\varepsilon}) + 2)\psi, \quad \text{for } |t + h| \le \rho_{\varepsilon}(r) \quad (6.10)$$

where

$$\varrho_{\varepsilon}(y) := \frac{\delta_0}{\varepsilon} \left(1 - \eta_{\varepsilon}(\varepsilon r) \right) + \eta_{\varepsilon}(\varepsilon r) \varepsilon^1 F_{\varepsilon}(\varepsilon r) - 2, \quad y = Y_{\varepsilon}(r, \Theta) \in \Sigma_{\varepsilon}$$

Now, we extend equation (6.10) to the whole $\Sigma_{\varepsilon} \times \mathbb{R}$. First, let us introduce the differential operator

$$B := \zeta_2 [\Delta_{\mathcal{N}_{\varepsilon,h}} - \partial_{tt} - \Delta_{\Sigma_{\varepsilon}}].$$

Recall that $\Delta_{\Sigma_{\varepsilon}}$ is nothing but the Laplace-Beltrami operator of Σ_{ε} and which in the local coordinates $Y_{\varepsilon,\pm}(r,\Theta)$, has the expression

$$\Delta_{\Sigma_{\varepsilon,\pm}} = \partial_{rr} + \frac{N-1}{r} \partial_r + \frac{1}{r^2} \Delta_{S^{N-1}} + \varepsilon^2 \mathcal{O}(\varepsilon^{-2} r^{-2}) \partial_r r + \varepsilon^3 \mathcal{O}(\varepsilon^{-3} r^{-3}) \partial_r r + \varepsilon^2 \mathcal{O}(\varepsilon^{-3} r^{-3}) \partial_r r + \varepsilon^3 \mathcal{O}(\varepsilon^{-3} r^{-3}) \partial_r r$$

Clearly, B vanishes in the domain

$$|t+h| \ge \frac{\delta_0}{\varepsilon} (1 - \eta_{\varepsilon}(\varepsilon r)) + \eta_{\varepsilon}(\varepsilon r) \varepsilon^1 F_{\varepsilon}(\varepsilon r) - 1, \quad y = Y_{\varepsilon}(r, \Theta) \in \Sigma_{\varepsilon}$$

and so, instead of equation (6.10), we consider the equation

$$\partial_{tt}\phi + \Delta_{\Sigma_{\varepsilon}}\phi + F'(w(t))\phi = -\widetilde{S}(\mathbf{w}_{\varepsilon}) - B(\phi)$$
$$- [F'(\zeta_{2}\mathbf{w}_{\varepsilon}) - F'(w(t))]\phi - \zeta_{2}(F'(\mathbf{w}_{\varepsilon}) + 2)\psi - \zeta_{2}N(\phi + \psi), \quad \text{in } M_{\alpha} \times \mathbb{R}$$
 (6.11)

and where we have denoted

$$\widetilde{S}(\mathbf{w}_{\varepsilon}) = -\varepsilon^{2} \left(\Delta_{\Sigma} h + |A_{\Sigma}|^{2} h \right) w'(t) - 6 \left(1 - w^{2}(t) \right) e^{\sqrt{2}t} \chi_{\varepsilon} 2\sqrt{2} e^{-2\sqrt{2}\varepsilon^{-1}F_{\varepsilon}} h$$

$$-\chi_{\varepsilon} E_0(\varepsilon r) w'(t) - \varepsilon^2 |A_{\Sigma}|^2 t \, w'(t) + g_0(t) e^{-2\sqrt{2}\varepsilon^{-1}F_{\varepsilon}} + \varepsilon^2 |\nabla_{\Sigma} h|^2 \, w''(t)$$

$$-2\sqrt{2}c_0a_0\chi_{\varepsilon}(\varepsilon r)e^{-2\sqrt{2}\varepsilon^{-1}F_{\varepsilon}}\left[e^{-2\sqrt{2}h}-1+2\sqrt{2}h\right]+\zeta_1R.$$

and where

$$|D_p R(\varepsilon r, t, p, q)| + |D_q R(\varepsilon r, t, p, q)| + |R(\varepsilon r', t, p, q)| \le C\varepsilon^{2-\tau} (1 + |\varepsilon r|)^{-2+\tau} e^{-\sigma|t|}$$
(6.12)

for any $0 < \sigma < \sqrt{2}$ and $\tau = \frac{\sigma}{\sqrt{2}}$.

Observe that $\widetilde{S}(\mathbf{w}_{\varepsilon})$ coincides with $S(\mathbf{w}_{\varepsilon})$ where $\zeta_1 = 1$, but we have basically cut-off the parts in $S(\mathbf{w}_{\varepsilon})$ that, in the local coordinates $X_{\varepsilon,\pm,h}$, are not defined for all $t \in \mathbb{R}$.

Using (5.16) and since the support of ζ_2 is contained in a region of the form

$$|t+h| \le \frac{C}{\varepsilon} (1-\eta_{\varepsilon}) + \frac{\eta_{\varepsilon}}{2\sqrt{2}} \left(\log(\frac{2\sqrt{2}}{\varepsilon^2}) + \log(\varepsilon^2 r^2) + \log(2(N-2)) \right)$$

we compute directly the size of this error to obtain that

$$\|\widetilde{S}(\mathbf{w}_{\varepsilon})\|_{p,2-\frac{\sigma}{\sqrt{2}},\sigma} \le C\varepsilon^{2-\frac{\sigma}{\sqrt{2}}} \tag{6.13}$$

for any $\sigma \in (0, \sqrt{2})$.

To solve system (6.9)-(6.11), we first solve equation (6.9), using the fact that the potential $2-(1-\zeta_3)(F'(\mathbf{w}_{\varepsilon})+2)$ is uniformly positive, so that the linear operator there behaves like $\Delta_{\mathbb{R}^{N+1}}-2$. A solution $\psi=\Psi(\phi)$ is then found using contraction mapping principle. We collect this discussion in the following proposition, that will be proven in detail in section 7.

Proposition 6.1. Assume $0 < \sigma < \sqrt{2}$, $\mu > 0$, p > 2 and let the function h satisfy (3.23). Then, for every $\varepsilon > 0$ sufficiently small and for a fixed function ϕ , satisfying that

$$\|\phi\|_{2,p,\mu,\sigma} \le 1$$

equation (6.9) has a unique solution $\psi = \Psi(\phi)$. Even more, the operator $\psi = \Psi(\phi)$ turns out to be Lipschitz in ϕ . More precisely, $\psi = \Psi(\phi)$ satisfies that

$$\|\psi\|_{X} := \|D^{2}\psi\|_{p,\hat{\mu},\sim} + \|(1+r^{\hat{\mu}}(\alpha x))D\psi\|_{L^{\infty}(\mathbb{R}^{N+1})} + \|(1+r^{\hat{\mu}}(\alpha x))\psi\|_{L^{\infty}(\mathbb{R}^{N+1})}$$

$$\leq C\left(\alpha^{2+\frac{\sigma}{\sqrt{2}}-\alpha} + \varepsilon^{\frac{\sigma}{\sqrt{2}}-\alpha}\|\phi\|_{2,p,\mu,\sigma}\right)$$
(6.14)

where $0 < \hat{\mu} < \min(2\mu, \mu + \varrho\sqrt{2}, 2 + \varrho\sqrt{2})$ and

$$\|\Psi(\phi) - \Psi(\hat{\phi})\|_{X} \le C\varepsilon^{\frac{\varrho}{\sqrt{2}} - \alpha} \|\phi - \hat{\phi}\|_{2, p, \mu, \sigma}. \tag{6.15}$$

Hence, using Proposition 6.1, we solve equation (6.11) with $\psi = \Psi(\phi)$. Let us set

$$\mathbf{N}(\phi) := B(\phi) + [F'(\zeta_2 \mathbf{w}_{\varepsilon}) - F'(w(t))]\phi + \zeta_2(F'(\mathbf{w}_{\varepsilon}) + 2)\Psi(\phi) + \zeta_2N[\phi + \Psi(\phi)].$$

We need to solve

$$\partial_{tt}\phi + \Delta_{\Sigma_{\varepsilon}}\phi + F'(w(t))\phi = -\widetilde{S}(w_{\varepsilon}) - \mathbf{N}(\Phi), \text{ in } \Sigma_{\varepsilon} \times \mathbb{R}.$$
 (6.16)

To solve system (6.16), we solve a nonlinear and nonlocal problem for ϕ , in such a way that we eliminate the parts of the error that do not contribute to the projections onto w'(t). This step can be though as an improvement of the approximation \mathbf{w}_{ε} .

We use the fact that the error has the size

$$\|\widetilde{S}(\mathbf{w}_{\varepsilon})\|_{p,2-\frac{\sigma}{\sqrt{2}},\sigma} \le \varepsilon^{2-\frac{\sigma}{\sqrt{2}}}$$
 (6.17)

and, as we will see in section 7, for $\tau = \frac{\sigma}{\sqrt{2}} > 0$, $\mathbf{N}(\phi)$ satisfies that

$$\|\mathbf{N}(\phi)\|_{p,4-\frac{\sigma}{\overline{G}},\sigma} \le C\varepsilon^{3-\tau} \tag{6.18}$$

$$\|\mathbf{N}(\phi_1) - \mathbf{N}(\phi_2)\|_{p,4-\frac{\sigma}{G},\sigma} \le C \varepsilon \|\phi_1 - \phi_2\|_{2,p,2,\sigma},\tag{6.19}$$

for $\phi_1, \phi_2 \in B_{\varepsilon}$ a ball of radius $\mathcal{O}(\varepsilon^{2-\frac{\sigma}{\sqrt{2}}})$ in the norm $\|\cdot\|_{2,p,2-\frac{\sigma}{\sqrt{2}},\sigma}$. A direct application of the contraction mapping principle allows us to solve the projected system

$$\partial_{tt}\phi + \Delta_{\Sigma_{\varepsilon}}\phi + F'(w(t))\phi = -\widetilde{S}(w_{\varepsilon}) - \mathbf{N}(\phi) + c(y)w'(t), \quad \text{in } \Sigma_{\varepsilon} \times \mathbb{R}.$$
 (6.20)

$$\int_{\mathbb{D}} \phi(y, t) w'(t) dt = 0, \quad y \in \Sigma_{\varepsilon}.$$
(6.21)

where

$$c(y) = \int_{\mathbb{D}} \left[\widetilde{S}(\mathbf{w}_{\varepsilon}) + \mathbf{N}(\phi) \right] w'(t) dt.$$

This solution ϕ , defines a Lipschitz operator $\phi = \Phi(h)$. This information is collected in the following proposition.

Proposition 6.2. Assume $0 < \mu \le 2$, $0 < \sigma < \sqrt{2}$ and p > 2. For every $\varepsilon > 0$ small enough, there exists an universal constant C > 0, such that system (6.20)-(6.21) has a unique solution $\phi = \Phi(h)$, satisfying

$$\|\Phi\|_{2,p,2-\frac{\sigma}{\sqrt{2}},\sigma} \leq C\varepsilon^{2-\frac{\sigma}{\sqrt{2}}}$$

and

$$\|\Phi(h) - \Phi(\hat{h})\|_{2,p,2-\frac{\sigma}{\sqrt{2}},\sigma} \le C \,\varepsilon^{2-\tau} \|h - \hat{h}\|_{2,p,2+\beta}$$

for some fixed $\beta > 0$ small.

At point we remark that to complete the proof of Theorem 1, it remains to adjust the nodal set, using the parameter function h, in such a way that the coefficient c(y) in the projected equation (6.20) becomes identically zero. This task is pursued in the next subsection.

6.2 Adjusting the nodal sets. First, we estimate the size of the error of the projected problem. For that we estate the following Lemma.

Lemma 6.1. Assume $\Psi(y,t)$ is a function defined in $\Sigma_{\varepsilon} \times \mathbb{R}$ and for which

$$\sup_{(y,t)\in\Sigma_{\varepsilon}\times\mathbb{R}} (1+r(\varepsilon y)^{\mu})e^{\sigma|t|} \|\Psi\|_{L^{p}(B_{1}(y,t))} < \infty$$

for some $\sigma, \mu > 0$ and p > 2. The function defined in Σ as

$$g(y) := \int_{\mathbb{R}} \Psi\left(\frac{y}{\varepsilon}, t\right) w'(t) dt$$

satisfies the estimate

$$||g||_{p,\mu} \le C\varepsilon^{-\frac{N}{p}} \sup_{(y,t)\in\Sigma_{\varepsilon}\times\mathbb{R}} (1+r(y)^{\mu})e^{\sigma|t|} ||\Psi||_{L^{p}(B_{1}(y,t))}.$$

Proof. It is enough to notice that

$$\int_{R-1<|y|$$

$$\leq C\varepsilon^{N} \sum_{j\geq N_{0,\varepsilon}}^{N_{1,\varepsilon}} \sum_{|k|\geq 0} \int_{j-1<|\hat{y}|< j+1} \int_{|t-k|<1} |\Psi(\hat{y},t)|^{p} |w'(t)|^{p} dt dV_{\Sigma_{\varepsilon}}$$

where $N_{0,\varepsilon} := \lfloor \frac{R-1}{\varepsilon} \rfloor$ and $N_{1,\varepsilon} := \lceil \frac{R+1}{\varepsilon} \rceil$. Since the set $\{\hat{y} : j-1 < |\hat{y}| < j+1\}$ can be cover with $\mathcal{O}(j^{-N})$ balls of radius 1, we conclude that

$$\int_{R-1<|y|< R+1} |g(y)|^p dV_{\Sigma} \leq C \varepsilon^{N-\mu p} \|\Psi\|_{p,\mu,\sigma}^p \sum_{j\geq N_{0,\varepsilon}}^{N_{1,\varepsilon}} \sum_{|k|\geq 0} e^{-\sigma p|k|} |j|^{-\mu p+N}$$

$$\leq C \varepsilon^{N-\mu p} \|\Psi\|_{p,\mu,\sigma}^p \sum_{j=N_{0,\varepsilon}}^{N_{1,\varepsilon}} \int_{j<|\hat{y}|< j+1} |\hat{y}|^{-\mu p+1} dV_{\Sigma_{\varepsilon}}$$

$$\leq C \varepsilon^{N-\mu p} \|\Psi\|_{p,\mu,\sigma}^p \int_{\frac{R-1}{\varepsilon} <|\hat{y}|< \frac{R+1}{\varepsilon}} |\hat{y}|^{1-\mu p} dV_{\Sigma_{\varepsilon}}$$

so that

$$||g||_{p,\beta} \le C\varepsilon^{-\frac{1}{p}}||\Psi||_{p,\mu,\sigma}$$

for $\beta \leq \mu - \frac{N}{p}$.

To conclude the proof of Theorem 1, we choose the function h in such a way that

$$c(y) = \int_{\mathbb{R}} \left[\widetilde{S}(\mathbf{w}_{\varepsilon}) + \mathbf{N}(\Phi) \right] w'(t) dt = 0, \quad y \in \Sigma_{\varepsilon}.$$

Let us write

$$\mathbf{Q}_1(h) := \varepsilon^{-2} \int_{\mathbb{R}} \widetilde{S}(\mathbf{w}_{\varepsilon}) w'(t) dt, \quad \mathbf{Q}_2(h) := \varepsilon^{-2} \int_{\mathbb{R}} \mathbf{N}(\Phi) w'(t) dt.$$

From (6.18), (6.19) and lemma 6.1 we find for $\tau = \frac{\sigma}{\sqrt{2}}$ that

$$\|\mathbf{Q}_2(h)\|_{p,4-\tau-\frac{N}{p}} \leq C\varepsilon^{1+\tau-\frac{1}{p}}$$

and

$$\|\mathbf{Q}_2(h) - \mathbf{Q}_2(\hat{h})\|_{p,4-\frac{N}{p}} \le C\varepsilon^{1+\tau-\frac{1}{p}} \|h - \hat{h}\|_{2,p,2+\beta}.$$

Next, we analyze the term $Q_1(h)$. Using the decomposition in (5.7) we set

$$c_0 = \|w'\|_{L^2(\mathbb{R})}^2, \quad a_0 = \|w'\|_{L^2}^{-2} \int_{\mathbb{R}} 6(1 - w^2(t))e^{\sqrt{2}t}w'(t)dt$$

and evaluating at $\varepsilon^{-1}y$ we find that

$$\varepsilon^{2} Q_{1}(h) = \int_{\mathbb{R}} \widetilde{S}(\mathbf{w}_{\varepsilon}) w'(t) dt =$$

$$-c_{0} \varepsilon^{2} \left(\Delta_{\Sigma} h + |A_{\Sigma}|^{2} h \right) + 2\sqrt{2} a_{0} e^{\sqrt{2}t} \chi_{\varepsilon} e^{-2\sqrt{2}\varepsilon^{-1}F_{\varepsilon}} h - c_{0} \chi_{\varepsilon} E_{0}(r) w'(t)$$

$$-2\sqrt{2} c_{0} a_{0} \chi_{\varepsilon}(\varepsilon r) e^{-2\sqrt{2}\varepsilon^{-1}F_{\varepsilon}} \left[e^{-2\sqrt{2}h} - 1 + 2\sqrt{2}h \right] + \int_{\mathbb{R}} \zeta_{1} R w'(t) dt.$$

We know from (5.14) that

$$||E_0||_{p,4} \le C \, \varepsilon^4$$

On the other hand, if h is such that $||h||_{2,p,2+\beta} \leq C\varepsilon^{\tau}$, then

$$\left| 2\sqrt{2}c_0 \, a_0 \chi_{\varepsilon}(\varepsilon r) \, e^{-2\sqrt{2}\varepsilon^{-1}F_{\varepsilon}} \, \left[\, e^{-2\sqrt{2}h} \, - \, 1 \, + 2\sqrt{2} \, h \right] \right| \le C\varepsilon^2 (1+r)^{-2}h^2$$

so that

$$\|-2\sqrt{2}c_0 a_0 \chi_{\varepsilon} e^{-2\sqrt{2}\varepsilon^{-1}F_{\varepsilon}} \left[e^{-2\sqrt{2}h} - 1 + 2\sqrt{2}h \right] \|_{p,2+\beta} \le C\varepsilon^{2\tau}, \quad \tau > 0.$$

and the Lipschitz dependence of this term respect to h and in the topology described above, holds true with Lipschitz constant $\mathcal{O}(\varepsilon^{2+\tau})$.

Finally, from (6.12) it is straight forward to check that

$$\left\| \int_{\mathbb{R}} \zeta_1 Rw'(t) dt \right\|_{p,2+\beta} \le C\varepsilon^{2+\tau_1}, \quad \tau_1 > \tau$$

being as well a contraction, with Lipschitz constant $\mathcal{O}(\varepsilon^{2+\tau_1})$.

Hence applying Proposition (4.1) and a fixed point argument for h in the ball

$$B := \{ h \in W_{loc}^{2,p}(\Sigma) : ||h||_{2,p,2+\beta} \le C\varepsilon^{\tau} \}, \quad p > N$$

we find h in such a way that c(y) = 0, and this finishes the proof of Theorem 1. The next section is devoted to provide the proofs of the propositions and lemmas mentioned here.

7. Gluing reduction and solution to the projected problem.

In this section, we prove propositions 6.1 and 6.2. The notations we use in this section have been set up in sections 4 and 5.

7.1 Solving the Gluing System. Given a fixed function ϕ such that $\|\phi\|_{2,p,\mu,\sigma} \leq 1$, we solve problem (6.9). To begin with, we observe that setting

$$Q_{\varepsilon}(x) = 2 - (1 - \zeta_2) \left[F'(\mathbf{w}_{\varepsilon}) + 2 \right].$$

there exist constants a < b, independent of ε , such that

$$0 < a \le Q_{\varepsilon}(x) \le b$$
, for every $x \in \mathbb{R}^{N+1}$.

Using this remark, we study the problem

$$\Delta \psi - Q_{\varepsilon}(x)\psi = g(x), \quad x \in \mathbb{R}^{N+1} \tag{7.1}$$

for a given g = g(x) such that

$$||g||_{p,\hat{\mu},\sim} := \sup_{x \in \mathbb{R}^{N+1}} (1 + R^{\hat{\mu}}(\varepsilon x)) ||g||_{L^p(B_1(x))}.$$

Solvability theory for equation (7.1) is collected in the following lemma whose proof follows the same lines as in lemma 7.1 in [13] and [16].

Lemma 7.1. Assume p > 2 and $\hat{\mu} \ge 0$. There exists a constant C > 0 and $\varepsilon_0 > 0$ small enough such that for $0 < \varepsilon < \varepsilon_0$ and any given g = g(x) with $||g||_{p,\hat{\mu},\sim} < \infty$, equation (7.1) has a unique solution $\psi = \psi(g)$, satisfying the a-priori estimate

$$\|\psi\|_X \leq C\|g\|_{p,\hat{\mu},\sim}$$

where

$$\|\psi\|_{X} := \|D^{2}\psi\|_{p,\hat{\mu},\sim} + \|(1+r(\alpha x)^{\hat{\mu}}(x))D\psi\|_{L^{\infty}(\mathbb{R}^{N+1})} + \|(1+r^{\hat{\mu}}(\alpha x))\psi\|_{L^{\infty}(\mathbb{R}^{N+1})}.$$

Now we prove Proposition 6.1. Denote by X, the space of functions $\psi \in W^{2,p}_{loc}(\mathbb{R}^{N+1})$ such that $\|\psi\|_X < \infty$ and let us denote by $\Gamma(g) = \psi$ the solution to the equation (7.1) from the previous lemma. We see that the linear map Γ is continuous i.e

$$\|\Gamma(g)\|_X \leq C\|g\|_{p,\hat{\mu},\sim}$$

with $0 < \hat{\mu} < \min(2\mu, \mu + \varrho\sqrt{2}, 2 + \varrho\sqrt{2})$. Using this we can recast (6.9) as a fixed point problem, in the following manner

$$\psi = -\Gamma \left((1 - \zeta_2) S(\mathbf{w}_{\varepsilon}) + \nabla \zeta_2 \cdot \nabla \phi + \phi \Delta \zeta_2 + (1 - \zeta_2) N \left[\zeta_3 \phi + \psi \right] \right) \tag{7.2}$$

Under conditions (3.23) and $\|\phi\|_{2,p,\mu,\sigma} \le 1$, we estimate the size of the right-hand side in (7.2). Also recall from (6.13)

$$\|\widetilde{S}(\mathbf{w}_{\varepsilon})\|_{p,2-\frac{\sigma}{\sqrt{2}},\sigma} \leq C\varepsilon^{2-\frac{\sigma}{\sqrt{2}}}$$

for any $\sigma \in (0, \sqrt{2})$. Since the support of ζ_2 is contained in a region of the form

$$\varrho_{\varepsilon} - 1 \le |t + h| \le \varrho_{\varepsilon} + 1$$

where

$$\varrho_{\varepsilon}(x) = \frac{\delta_0}{\varepsilon} \left(1 - \eta_{\varepsilon}(\varepsilon r) \right) + \eta_{\varepsilon}(\varepsilon r) \varepsilon^1 F_{\varepsilon}(\varepsilon r) - 1, \quad y = Y_{\varepsilon}(r, \Theta) \in \Sigma_{\varepsilon}$$

we estimate directly for $0 < \hat{\mu} \le 2$, to get that

$$\|(1-\zeta_2)S(\mathbf{w}_{\varepsilon})\|_{n,\hat{\mu},\sim} \leq C\varepsilon^2$$

for some $\varepsilon > 0$ sufficiently small.

As for the second term in the right-hand side of (7.2), the following holds true

$$|2\nabla\zeta_{2}\cdot\nabla\phi + \phi\Delta\zeta_{2}| \leq C(1-\zeta_{2})(1+r^{\mu}(\varepsilon y))^{-1}e^{-\sigma|t|}\|\phi\|_{2,p,\mu,\sigma}$$

$$\leq C\varepsilon^{\frac{\sigma}{\sqrt{2}}}(1+r^{\mu+\frac{\sigma}{\sqrt{2}}}(\varepsilon r))^{-1}\|\phi\|_{2,p,\mu,\sigma}.$$

This implies that

$$||2\nabla \zeta_2 \cdot \nabla \phi + \phi \Delta \zeta_2||_{p,\mu + \frac{\sigma}{\sqrt{2}}, \sim} \le C \varepsilon^{\frac{\sigma}{\sqrt{2}}} ||\phi||_{2,p,\mu,\sigma}.$$

Finally we must check the lipschitz character of $(1-\zeta_2)N[\zeta_2\phi+\psi]$. Take $\psi_1,\psi_2\in X$. Then

$$(1 - \zeta_2) |N [\zeta_2 \phi + \psi_1] - N [\zeta_2 \phi + \psi_2]| \le$$

$$\le (1 - \zeta_2) |F(\mathbf{w} + \zeta_1 \phi + \psi_1) - F(\mathbf{w} + \zeta_1 \phi + \psi_2) - F'(\mathbf{w})(\psi_1 - \psi_2)|$$

$$\le C (1 - \zeta_2) \sup_{s \in [0,1]} |\zeta_1 \phi + s\psi_1 + (1 - s)\psi_2| |\psi_1 - \psi_2|.$$

So, we see that

$$\begin{split} &\| (1 - \zeta_2) \, N \, [\zeta_2 \phi + \psi_1] - (1 - \zeta_2) \, N \, [\zeta_2 \phi + \psi_2] \|_{p,2\hat{\mu},\sim} \\ &\leq C \varepsilon^{\frac{\sigma}{\sqrt{2}}} \, (\|\phi\|_{\infty,\mu,\sigma} + \|\psi_1\|_X + \|\psi_2\|_X) \, \|\psi_1 - \psi_2\|_{\infty,\hat{\mu},\sim} \\ &\leq C \varepsilon^{\frac{\sigma}{\sqrt{2}}} \|\psi_1 - \psi_2\|_{\infty,\hat{\mu},\sim}. \end{split}$$

In particular, we take advantage of the fact that $N(\varphi) \sim \varphi^2$, to find that there exists $\alpha > 0$ small such that

$$\|(1-\zeta_2) N(\zeta_2\phi)\|_{p,2\mu,\sim} \leq C\varepsilon^{\frac{\sigma}{\sqrt{2}}-\alpha} \|\phi\|_{2,p,\mu,\sigma}^2.$$

Consider $\widetilde{\Gamma}: X \to X$, $\widetilde{\Gamma} = \widetilde{\Gamma}(\psi)$ the operator given by the right-hand side of (7.2). From the previous remarks we have that $\widetilde{\Gamma}$ is a contraction provided $\varepsilon > 0$ is small enough and so we have found $\psi = \widetilde{\Gamma}(\psi)$ the solution to (6.9) with

$$\|\psi\|_X \le C\left(\varepsilon^2 + \varepsilon^{\frac{\sigma}{\sqrt{2}}} \|\phi\|_{2,p,\mu,\sigma}\right).$$

We can check directly that $\Psi(\phi) = \psi$ is Lipschitz in ϕ , i.e

$$\begin{split} \|\Psi(\phi_1) - \Psi(\phi_2)\|_X &\leq \\ C \, \|(1 - \zeta_2) \, [N \, (\zeta_2 \phi_1 + \Psi(\Phi_1)) - N \, (\zeta_2 \phi_2 + \Psi(\phi_2))] \|_{p,2\mu,\sim} \\ &\leq C \varepsilon^{\frac{\sigma}{\sqrt{2}}} \, (\|\Psi(\phi_1) - \Psi(\phi_2)\|_X + \|\phi_1 - \phi_2\|_{2,\nu,\mu,\sigma}) \end{split}$$

so that for ε small, we conclude

$$\|\Psi(\phi_1) - \Psi(\phi_2)\|_X \le C\varepsilon^{\tau} \|\phi_1 - \phi_2\|_{2,p,\mu,\sigma}.$$

7.2 Solving the Projected equation (6.20)-(6.21). Now we solve the projected problem

$$\partial_{tt}\phi + \Delta_{\Sigma_{\varepsilon}}\phi + F'(w(t))\phi = -\widetilde{S}(w) - \mathbf{N}(\phi) + c(y)w'(t), \quad \text{in } \Sigma_{\varepsilon} \times \mathbb{R}.$$

$$\int_{\mathbb{R}} \phi(y,t)w'(t)dt = 0.$$

To do so, we need to study solvability for the linear equation

$$\partial_{tt}\phi + \Delta_{\Sigma_{\varepsilon}}\phi + F'(w(t))\phi = g(y,t) + c(y)w'(t), \quad \text{in } \Sigma_{\varepsilon} \times \mathbb{R}$$
 (7.3)

$$\int_{\mathbb{R}} \phi(y,t)w'(t)dt = 0. \tag{7.4}$$

Solvability of (7.3)-(7.4) is based upon the fact that the heteroclinic solution w(t) is nondegenerate in the sense, that the following property holds true.

Lemma 7.2. Assume that $\phi \in L^{\infty}(\mathbb{R}^{N+1})$ and assume $\phi = \phi(x',t)$ satisfies

$$L(\phi) := \partial_{tt}\phi + \Delta_{\mathbb{R}^N}\phi + F'(w(t))\phi = 0, \quad \text{in } \mathbb{R}^N \times \mathbb{R}.$$
 (7.5)

Then $\phi(x',t) = C w'(t)$, for some constant $C \in \mathbb{R}$.

For the detailed proof of this lemma we refer the reader to [13], [16] and references therein.

The linear theory we need to solve system (6.21), is collected in the following proposition, whose proof is again contained in essence in proposition 4.1 in [13] and [16].

Proposition 7.1. Assume p > 2, $0 < \sigma < \sqrt{2}$ and $\mu \ge 0$. There exist C > 0, an universal constant, and $\varepsilon_0 > 0$ small such that, for every $\varepsilon \in (0, \varepsilon_0)$ and any given g with $\|g\|_{p,\mu,\sigma} < \infty$, problem (7.3)-(7.4) has a unique solution (ϕ, c) with $\|\phi\|_{p,\mu,\sigma} < \infty$, satisfying the apriori estimate

$$||D^2\phi||_{p,\mu,\sigma} + ||D\phi||_{\infty,\mu,\sigma} + ||\phi||_{\infty,\mu,\sigma} \le C||g||_{p,\mu,\sigma}.$$

Using Proposition 7.1, we are ready to solve system (6.20)-(6.21). First, recall that as stated in (6.13)

$$\|\widetilde{S}(\mathbf{w}_{\varepsilon})\|_{p,2-\frac{\sigma}{\sqrt{2}},\sigma} \le C\varepsilon^{2-\frac{\sigma}{\sqrt{2}}}.$$
(7.6)

From proposition 6.1 we have a nonlocal operator $\psi = \Psi(\phi)$. We want to solve the following problem Recall that

$$\mathbf{N}(\phi) := B(\phi) + \left[F'(\zeta_2 \mathbf{w}_{\varepsilon}) - F'(w(t)) \right] \phi +$$

$$+\zeta_2[F'(\mathbf{w}_{\varepsilon})+2]\Psi(\phi)+\zeta_2N(\phi+\Psi(\phi)).$$

Let us denote

$$N_1(\phi) := B(\phi) + [F'(\zeta_2 \mathbf{w}_{\varepsilon}) - F'(w(t))] \phi$$

$$N_2(\phi) := \zeta_2 \left[F'(\mathbf{w}_{\varepsilon}) + 2 \right] \Psi(\phi)$$

$$N_3(\phi) := \zeta_2 N(\phi + \Psi(\phi)).$$

We need to investigate the Lipschitz character of N_i , i = 1, 2, 3. We begin with N_3 . Observe that

$$|N_3(\phi_1) - N_3(\phi_2)| = \zeta_2 |N(\phi_1 + \Psi(\phi_1)) - N(\phi_2 + \Psi(\phi_2))|$$

$$\leq C\zeta_2 \sup_{\tau \in [0,1]} |\tau(\phi_1 + \Psi(\Phi_1)) + (1-\tau)(\phi_2 + \Psi(\phi_2))| \cdot |\phi_1 - \phi_2 + \Psi(\phi_1) - \Psi(\phi_2)|$$

$$\leq C \left[|\Psi(\Phi_2)| + |\phi_1 - \phi_2| + |\Psi(\phi_1) - \Psi(\phi_2)| + |\phi_2| \right] \cdot \left[|\phi_1 - \phi_2| + |\Psi(\phi_1) - \Psi(\phi_2)| \right].$$

This implies that

$$||N_3(\phi_1) - N_3(\phi_2)||_{p,2\mu,\sigma} \le$$

$$\le C \left[\varepsilon^2 + ||\phi_1||_{\infty,\mu,\sigma} + ||\phi_2||_{\infty,\mu,\sigma} \right] \cdot ||\phi_1 - \phi_2||_{\infty,\mu,\sigma} .$$

Now we check on $N_1(\phi)$. Clearly, we just have to pay attention to $B(\phi)$. But notice that $B(\phi)$ is linear on ϕ and

$$B(\phi) = -\varepsilon^2 \left\{ \Delta_{\Sigma} h + |A_{\Sigma}|^2 (t+h) \right\} \partial_t \phi$$
$$-2\varepsilon \nabla_{\Sigma} h(\varepsilon y) \partial_t \nabla_{\Sigma_{\varepsilon}} \phi + \varepsilon^2 [\nabla h(\varepsilon y)]^2 \partial_{tt} \phi + D_{\varepsilon,h}(\phi),$$

where the differential operator $D_{\varepsilon,h}$ is given in (3.25). From assumptions (3.23) made on the function h, we have that

$$||N_1(\phi_1) - N_1(\phi_2)||_{p,1+\mu,\sigma} \le C\varepsilon ||\phi_1 - \phi_2||_{2,p,\mu,\sigma}.$$

Then, assuming that $\|\phi\|_{2,p,\mu,\sigma} \leq A\varepsilon^{2-\frac{\sigma}{\sqrt{2}}}$, we have that

$$\|\mathbf{N}(\phi)\|_{p,1+\mu,\sigma} \le C\varepsilon^{3-\frac{\sigma}{\sqrt{2}}}$$

Setting $T(g) = \phi$ the linear operator given by the Lemma 7.1, we recast problem (6.20) as the fixed point problem

$$\phi = T(-\widetilde{S}(\mathbf{w}_{\varepsilon}) - \mathbf{N}(\phi)) =: \mathcal{T}(\phi)$$

in the ball

$$B_{\varepsilon} := \left\{ \phi \in W_{loc}^{2,p}(\Sigma_{\varepsilon} \times \mathbb{R}) : \|\phi\|_{2,p,\mu,\sigma} \le A \, \varepsilon^{2 - \frac{\sigma}{\sqrt{2}}} \right\}$$

where $0 < \mu \le 2 - \frac{\sigma}{\sqrt{2}}$.

Observe that

$$\|\mathcal{T}(\phi_1) - \mathcal{T}(\phi_2)\|_{2,p,\mu+1,\sigma} \le C \|\mathbf{N}(\phi_1) - \mathbf{N}(\phi_2)\|_{p,\mu+1,\sigma} \le C \varepsilon \|\phi_1 - \phi_2\|_{2,p,\mu,\sigma}, \quad \Phi_1, \phi_2 \in B_{\varepsilon}.$$

On the other hand, because C and K_1 are universal constants and taking A large enough independent of $\varepsilon > 0$, we have that

$$\|\mathcal{T}(\phi)\|_{2,p,\mu,\sigma} \leq C \left(\|\widetilde{S}(\mathbf{w}_{\varepsilon})\|_{p,2-\frac{\sigma}{\sqrt{2}},\sigma} + \|\mathbf{N}(\phi)\|_{p,4,\sigma} \right) \leq A \varepsilon^{2-\frac{\sigma}{\sqrt{2}}}, \quad \phi \in B_{\varepsilon}.$$

Hence, the mapping \mathcal{T} is a contraction from the ball B_{ε} onto itself. From the contraction mapping principle we get a unique solution

$$\phi = \Phi(h)$$

as required. As for the Lipschitz character of $\Phi(h)$ it comes from a lengthy by direct computation from the fact that

$$\|\Phi(h) - \Phi(\widetilde{h})\|_{2,p,2-\frac{\sigma}{\sqrt{2}},\sigma}$$

$$\leq C\|\widetilde{S}(\mathbf{w}_{\varepsilon},h) - \widetilde{S}(\mathbf{w}_{\varepsilon},\widetilde{h})\|_{p,2-\frac{\sigma}{\sqrt{2}},\sigma} +$$

$$+ \|N(\Phi(h)) - N(\Phi(\widetilde{h})\|_{p,3-\frac{\sigma}{\sqrt{2}},\sigma}.$$

We left to the reader to check on the details of the proof of the following estimate

$$\|\Phi(h) - \Phi(\widetilde{h})\|_{2,p,2-\frac{\sigma}{\sqrt{2}}} \sigma \sqrt{2} \le C\varepsilon^{2-\frac{\sigma}{\sqrt{2}}} \|h - \widetilde{h}\|_{p,\beta}$$

for h and h satisfying (3.23). This completes the proof of proposition 6.2 and consequently the proof of Theorem 1.

8. The proof of Theorem 2

8.1. Remarks on the profile of the solutions found in Theorem 1: In order to proof our second result, we need further information about the asymptotic profile of the solutions predicted in Theorem 1, as the parameter $\varepsilon > 0$ approaches zero.

Let us focus for the moment in the set $\Sigma_{\varepsilon,+}$. Taking a closer look to the error (6.12) of the initial approximation \bar{u} , described in section 5, we observe that the

$$S(\bar{u}) = -\varepsilon^{2} \left\{ \Delta_{\Sigma} h + |A_{\Sigma}|^{2} h \right\} w'(t) - 6 \left(1 - w^{2}(t) \right) e^{\sqrt{2}t} \chi_{\varepsilon} 2\sqrt{2} e^{-2\sqrt{2}\varepsilon^{-1}F_{\varepsilon}} h$$

$$-\chi_{\varepsilon} E_{0}(\varepsilon r) w'(t) + g_{0}(t) e^{-2\sqrt{2}\varepsilon^{-1}F_{\varepsilon}} - \varepsilon^{2} |A_{\Sigma}|^{2} t \, w'(t) + \varepsilon^{2} |\nabla_{\Sigma} h|^{2} w''(t)$$

$$-6(1 - w^{2}(t)) e^{\sqrt{2}t} \chi_{\varepsilon}(\varepsilon r) e^{-2\sqrt{2}\varepsilon^{-1}F_{\varepsilon}} \left[e^{-2\sqrt{2}h} - 1 + 2\sqrt{2}h \right] + R.$$

and where

$$|D_p R(\varepsilon r, t, p, q)| + |D_q R(\varepsilon r, t, p, q)| + |R(\varepsilon r', t, p, q)| \le C\varepsilon^{2-\tau} (1 + |\varepsilon r|)^{-2+\tau} e^{-\sigma|t|}$$

for some $0 < \sigma < \sqrt{2}$ and some $\tau = \frac{\sigma}{\sqrt{2}}$.

Let us consider $\psi_0(t)$ to be the bounded solution to the equation

$$\partial_{tt}\psi_0(t) + F'(w(t))\psi_0(t) = g_0(t), \quad t \in \mathbb{R}$$
(8.1)

given explicitly by the variations of parameters formula

$$\psi_0(t) = w'(t) \int_0^t w'(s)^{-2} \int_s^\infty w'(\xi) g_0(\xi) \, d\xi \, ds. \tag{8.2}$$

From (8.2), we obtain the estimate

$$\left\| \left(1 + e^{2\sqrt{2}t} \chi_{\{t>0\}} \right) \partial_t^{(j)} \psi_0 \right\|_{L^{\infty}(\mathbb{R})} \le C_j, \quad j \in \mathbb{N}.$$

Let us also consider the function $\psi_1(t)$ solving

$$\partial_{tt}\psi_1(t) + F'(w(t))\psi_1(t) = tw'(t), \quad t \in \mathbb{R}. \tag{8.3}$$

Proceeding as before, we see that

$$\psi_1(t) = -w(t) \int_0^t w'(s)^{-2} \int_s^\infty \xi w'(\xi)^2 d\xi ds$$

and $\psi_1(t) = -\frac{1}{2}tw'(t)$, from where the following estimate follows at once

$$||e^{\sigma|t|}\partial_t^{(j)}\psi_1||_{L^{\infty}(\mathbb{R})} \le C_j, \quad j \in \mathbb{N}, \quad 0 < \sigma < \sqrt{2}.$$

So, we take the function

$$\bar{\mathbf{w}}_{\varepsilon}(x) = \mathbf{w}_{\varepsilon} + \phi_0 \tag{8.4}$$

as a second approximation, in a way that in the region $\mathcal{N}_{\varepsilon,h}$ and in the coordinates $X_{\varepsilon,h}$

$$\phi_0(y,t) = -e^{-2\sqrt{2}\varepsilon^{-1}F_{\varepsilon}} \,\psi_0(t) \,+\, \varepsilon^2 |A_{\Sigma}(\varepsilon y)|^2 \,\psi_1(t)$$

for the upper hemisphere, while in the lower part, the function ϕ_0 is described as

$$\phi_0(y,t) = -e^{-2\sqrt{2}\varepsilon^{-1}F_{\varepsilon}} \psi_0(-t) + \varepsilon^2 |A_{\Sigma}(\varepsilon y)|^2 \psi_1(t).$$

Proceeding as we did to verify (5.16) we verify that the error created by $\bar{\mathbf{w}}_{\varepsilon}$ has the size

$$||S(\bar{\mathbf{w}}_{\varepsilon})||_{p,2,\sigma} \le C\varepsilon^{2+\tau}$$

and we can run the reduction procedure in the exact fashion as we did in section 6 to find that our solutions u_{ε} having in the region $\mathcal{N}_{\varepsilon,h}$ the asymptotic profile

$$u_{\varepsilon} = \bar{u} + \phi_0 + \phi_{\varepsilon}$$

for a function ϕ_{ε} such that

$$\|\phi_{\varepsilon}\|_{2,p,2,\sigma} \le C\varepsilon^{2+\tau}, \quad \text{for } 0 < \tau < \sqrt{2}.$$

8.2. **Energy and Morse Index estimates:** Estimate the Morse Index of the solutions described in 1 are deeply connected with the number of negative eigenvalues for the linear Hardy equation

$$\Delta_{\mathbb{R}^N} \mathbf{k} + (2(N-2) + \lambda)r^{-2}\mathbf{k} = 0, \quad \text{in } \{R_{\varepsilon} < r < \infty\}, \quad \mathbf{k} \in L^{\infty}(\{r > R_{\varepsilon}\}). \tag{8.5}$$

Our analysis is based upon the fact that test functions of the form

$$v(x', x_{N+1}) \approx k(y) \left[w'(t) - w(t - 2\varepsilon^{-1}F_{\varepsilon} - 2h) \right], \text{ for } r(\varepsilon y) > R_{\varepsilon}.$$

with $k(y) = k(\varepsilon y)$, for $y \in \Sigma_{\varepsilon}$ and k as in (8.5), are negative direction for the quadratic form

$$Q(v,v) := \iint_{W_{\mathcal{P}}} |\nabla v|^2 - F'(u_{\varepsilon})v^2 dx$$

Due to the symmetries considered in our discussion, we focus our analysis in the upper part of the surface Σ_{ε} . The exact same analysis and computations hold true in the lower part of the dilated surface.

Let k be a smooth, axially symmetric and compactly supported function defined in Σ_{ε} whose support is contained in the set $\{r(\varepsilon y) > R_{\varepsilon}\}$. In the set $\Sigma_{\varepsilon,h,+}$ where $z = t + h(\varepsilon y)$, we consider a test function of the form

$$v(x', x_{N+1}) = k(y) \left[w'(t) - w(t - 2\varepsilon^{-1}F_{\varepsilon} - 2h) \right], \quad \text{for } r(\varepsilon y) > R_{\varepsilon}.$$
 (8.6)

To compute the euclidean Laplacian for v in $\mathcal{N}_{\varepsilon,h}$ we notice first from (8.4) that in the upper part of the dilated surface Σ_{ε} and in the coordinates $X_{\varepsilon,h}$

$$F'(u_{\varepsilon})w'(t) = F'(w)w' +$$

$$F''(w)w' \left[(-2e^{\sqrt{2}t} + \psi_0(t))e^{-2\sqrt{2}(\varepsilon^{-1}F_{\varepsilon} + h)} + \varepsilon^2 |A_{\Sigma}|^2 \psi_1(t) \right] + \mathcal{O}\left(\varepsilon^{2+\tau} (1 + r(\varepsilon y)^2)^{-1} e^{-\sigma|t|} \right). \quad (8.7)$$

As for the interaction term we compute

$$F'(u_{\varepsilon})w'(t-2\varepsilon^{-1}F_{\varepsilon}-2h)=F'(w(t-2\varepsilon^{-1}F_{\varepsilon}-2h))w'(t-2\varepsilon^{-1}F_{\varepsilon}-2h)$$

$$+\sqrt{2}(F'(w)-F'(1))e^{\sqrt{2}t}e^{-2\sqrt{2}(\varepsilon^{-1}F_\varepsilon+h)}+\mathcal{O}\left(\varepsilon^{2+\tau_1}e^{-\sigma}|t|(1+r(\varepsilon y))^{-2+\beta}\right)$$

for some $\beta > 0$.

Hence, from expression (3.24) and the above comments

$$\Delta_{X_{\varepsilon,h}} v + F'(u_{\varepsilon})v =$$

$$\underbrace{\Delta_{\Sigma_{\varepsilon}} k \, w' - \varepsilon^2 |A_{\Sigma}|^2 k \, t w'' + \varepsilon^2 |\nabla h|^2 k \, w''' + \varepsilon a_1 \, h \, \nabla_{\Sigma_{\varepsilon}} k \, w'}_{Q_1}$$

$$+ \underbrace{F''(w(t)) \left[(-2e^{\sqrt{2}t} - \psi_0(t))e^{-2\sqrt{2}(\varepsilon^{-1}F_{\varepsilon} + h)} + \varepsilon^2 |A_{\Sigma}|^2 \psi_1(t) \right] k \, w'}_{Q_2}$$

$$\underbrace{\sqrt{2}(F'(w) - F'(1))e^{\sqrt{2}t}e^{-2\sqrt{2}(\varepsilon^{-1}F_{\varepsilon} + h)} k}_{Q_2}$$

$$\underbrace{-w'' \left[\varepsilon^{2} \mathcal{J}_{\Sigma}(h)k + 2\varepsilon \nabla_{\Sigma} h \nabla_{\Sigma_{\varepsilon}} k + \varepsilon^{2} a_{1} h(\nabla_{\Sigma} h \nabla_{\varepsilon} k + \Delta_{\Sigma} h k) \right]}_{Q_{4}} \\
\underline{\varepsilon tw' \left[a_{1} \Delta_{\Sigma_{\varepsilon}} k + \varepsilon b_{1} \nabla_{\Sigma_{\varepsilon}} k \right]}_{Q_{5}} + \underbrace{\mathcal{O} \left(\varepsilon^{2+\tau_{1}} (1 + r(\varepsilon y)^{2+\tau_{1}})^{-1} e^{-\sigma|t|} \right)}_{Q_{6}} \tag{8.8}$$

for any $\tau_1 < \tau$.

Since F''(w) = -6w, taking derivatives in equations (8.1) and (8.3) and integrating against w'(t), we easily check that

$$\int_{\mathbb{R}} F''(w)(w')^{2} \psi_{2}(t) dt = -\int_{\mathbb{R}} (w'')^{2} dt$$

$$\int_{\mathbb{R}} F''(w)(w')^{2} \psi_{1}(t) dt = -\int_{\mathbb{R}} tw'w'' dt = \frac{1}{2} \int_{\mathbb{R}} (w')^{2} dt$$

$$\int_{\mathbb{R}} F''(w)(w')^{2} \left(-2e^{\sqrt{2}t} + \psi_{0}(t)\right) dt = \sqrt{2} \int_{\mathbb{R}} 6(1 - w^{2})e^{\sqrt{2}t} w' dt = \sqrt{2} a_{0} \int_{\mathbb{R}} (w'(t))^{2} dt.$$

Also, $F'(w) - F'(1) = 6(1 - w^2)$, and we find that

$$\sqrt{2} \int_{\mathbb{R}} 6(1 - w^2(t)) e^{\sqrt{2}t} w'(t) dt = \sqrt{2} a_0 \int_{\mathbb{R}} (w'(t))^2 dt.$$

Observe also that

$$\int_{|t| < \rho_{\varepsilon}} Q_i w'(t) dt = \int_{\mathbb{R}} Q_i w'(t) dt + \mathcal{O}\left(\varepsilon^{2+\tau_1} (1 + r(\varepsilon y)^{2+\tau_1})^{-1}\right)$$

where

$$\rho_{\varepsilon}(y) := \frac{\delta_0}{\varepsilon} \left(1 - \eta_{\varepsilon} \right) - \frac{\eta_{\varepsilon}}{2\sqrt{2}} \left(\log \left(\frac{2\sqrt{2} a_0}{\varepsilon^2} \right) + 2 \log(\varepsilon r) \right).$$

We observe also that

$$\int_{\mathbb{R}} \sum_{i=1}^{5} Q_i w'(t) dt = \left(\Delta_{\Sigma_{\varepsilon}} k + \varepsilon^2 |A_{\Sigma}|^2 k + 2\sqrt{2} a_0 e^{-2\sqrt{2}(\varepsilon^{-1} F_{\varepsilon} + h)} k + \varepsilon a_1 h \partial_{ij} k \right) \int_{\mathbb{R}} w'(t)^2 dt$$

and

$$\int_{\mathbb{D}} Q_6 w'(t) dt = \mathcal{O}(\varepsilon^2 r(\varepsilon y)^{-2}) \partial_{ij} k + \mathcal{O}\left(\varepsilon^3 r(\varepsilon y)^{-3}\right) \nabla_i k$$

Combining these computations, we have that

$$\int_{|t| \le \rho_{\varepsilon}} (\Delta v + F'(u_{\varepsilon})v) w'(t) dt = \left(\Delta_{\Sigma_{\varepsilon}} k + \varepsilon^{2} |A_{\Sigma}|^{2} k + \varepsilon a_{1} h \partial_{ij} k \right) \int_{\mathbb{R}} w'(t)^{2} dt$$

$$+ 2\sqrt{2} a_{0} e^{-2\sqrt{2}(\varepsilon^{-1} F_{\varepsilon} + h)} k \int_{\mathbb{R}} w'(t)^{2} dt$$

$$+ \mathcal{O}(\varepsilon^{2} r(\varepsilon y)^{-2}) D_{\Sigma_{\varepsilon}}^{2} k + \mathcal{O}(\varepsilon^{3} r(\varepsilon y)^{-3}) \nabla_{\Sigma_{\varepsilon}} k + \mathcal{O}(\varepsilon^{2+\tau_{1}} r(\alpha y)^{-2-\beta}) k.$$

Regarding computations in the lower part of Σ_{ε} we find that

$$F'(u_{\varepsilon})w' = F'(w)w' + F''(w)w' \left[(2e^{-\sqrt{2}t} - \psi_0(-t))e^{-2\sqrt{2}(\varepsilon^{-1}F_{\varepsilon} + h)} + \varepsilon^2 |A_{\Sigma}|^2 \psi_2(t) \right] + \mathcal{O}\left(\varepsilon^{2+\tau_1} (1 + r(\varepsilon y)^2)^{-1} e^{-\sigma|t|}\right)$$
(8.9)

and the interaction term this time takes the form

$$F'(u_{\varepsilon})w'(t-2\varepsilon^{-1}F_{\varepsilon}-2h) = F'(w(t-2\varepsilon^{-1}F_{\varepsilon}-2h))w'(t-2\varepsilon^{-1}F_{\varepsilon}-2h)$$

$$-\sqrt{2}(F'(w)-F'(1))e^{-\sqrt{2}t}e^{-2\sqrt{2}(\varepsilon^{-1}F_{\varepsilon}+h)}+\mathcal{O}\left(\varepsilon^{2+\tau_1}e^{-\sigma}|t|(1+|\varepsilon y|)^{-2-\beta}\right).$$

Similar computations as the ones above, in the region

$$W_{R_{\varepsilon}} = \{ x \in \mathcal{N}_{\varepsilon} : r(\varepsilon x) > R_{\varepsilon} \}$$

lead to the expression

$$Q(v,v) = \iint_{W_{R_{\varepsilon}}} |\nabla v|^2 - F'(u_{\varepsilon})v^2 dx$$

$$= 2 \int_{\mathbb{R}} w'(t)^2 dt \int_{\Sigma_{\varepsilon}} |\nabla_{\Sigma_{\varepsilon}} k|^2 - \varepsilon^2 |A_{\Sigma}|^2 k^2 + 2\sqrt{2}a_0 e^{-2\sqrt{2}(\varepsilon^{-1}F_{\varepsilon} + h)} k^2 dV_{\Sigma_{\varepsilon}}$$

$$+ \mathcal{O}\left(\varepsilon \int_{\Sigma_{\varepsilon}} |\nabla_{\Sigma} k|^2 + \varepsilon^2 (1 + r(\varepsilon y))^{-2-\beta} k^2 dV_{\Sigma_{\varepsilon}}\right).$$

Finally, since

$$2\sqrt{2}a_0e^{-2\sqrt{2}(\varepsilon^{-1}F_{\varepsilon}+h)} = \frac{2(N-2)\varepsilon^2}{r^2} + \varepsilon^2\mathcal{O}(r^{-2}(w_{\varepsilon}+h))$$

we find that

$$2\sqrt{2}a_0e^{-\sqrt{2}(\varepsilon^{-1}F_{\varepsilon}+h)} = \varepsilon^2 \left(\frac{2(N-2)}{r^2}(1+p(r))\right)$$

where $p(r) = \mathcal{O}(r^{-\beta})$ is smooth and

$$\nabla_{\Sigma} = \nabla_{\mathbb{R}^N} + +\varepsilon \mathcal{O}(r^{-1})\partial_r.$$

$$c \varepsilon^{2} (1+r)^{-4} \le |A_{\Sigma}|^{2} \le C \varepsilon^{2} (1+r)^{-4}, \quad r > R_{\varepsilon}.$$

Consequently, we obtain

$$Q(v,v) = \gamma_0 \int_{r(\varepsilon y) > R_{\varepsilon}} |\nabla_{\mathbb{R}^N} k|^2 - \varepsilon^2 \frac{2(N-2)}{r^2} k^2 dr + \varepsilon^{\tau_1} \mathcal{O}\left(\int_{r(\varepsilon y) > R_{\varepsilon}} |\nabla_{\mathbb{R}^N} k|^2 + \varepsilon^2 (1+\varepsilon r)^{-2-\beta}) k^2\right). \tag{8.10}$$

Let k(y) be an axially symmetric solution of the equation

$$\Delta_{\mathbb{R}^N} \mathbf{k} + (2(N-2) + \lambda)r^{-2}\mathbf{k} = 0$$
, in $\{R_{\varepsilon} < r < \infty\}$

with

$$\lambda = \lambda_m = \frac{1}{2}\sqrt{4m^2 - (N-10)(N-2)}$$

so that we may write explicitely

$$k(r) = \cos(\lambda_m \log(r)) r^{-\frac{N-2}{2}}.$$

Letting k(y) be axially symmetric and defined in Σ_{ε} in a way that in coordinates $X_{\varepsilon,h}(r,\theta,t)$ it coincides with $k(\varepsilon r)$, but cut off at infinity.

With $v(x', x_{N+1})$ as in (8.6) and from (8.10), it follows that

$$Q(v,v) = \iint_{W_R} |\nabla v|^2 - F'(u_{\varepsilon})v^2 dx$$

$$= -2\gamma_0 \varepsilon^{-N+2} \left[m^2 \int_{r(y)>R_{\varepsilon}} r^{-2} \mathbf{k}^2 dr + \mathcal{O}\left(\varepsilon^{\tau_1} \int_{r(y)>R_{\varepsilon}} |\nabla_{\mathbb{R}^N} \mathbf{k}|^2 + (1+r)^{-2} \mathbf{k}^2\right) \right]$$

and we conclude that Q(v, v) < 0 for every $\varepsilon > 0$ small.

Since $3 \le N \le 9$ and since $m \in \mathbb{N}$ is arbitrary, for every $m \in \mathbb{N}$, we obtain a negative direction for Q and consequently a lower bound for the Morse Index of u_{ε} . Taking $m \to \infty$ we conclude the proof of Theorem 2

References

- [1] O. AGUDELO, M. DEL PINO, J. WEI, Multiple catenoidal end solutions to the Allen-Cahn Equation in \mathbb{R}^3 . Peprint 2013.
- S. Allen and J. W. Cahn, A microscopic theory for antiphase boundary motion and its application to antiphase domain coarsening, Acta. Metall. 27 (1979), 1084-1095.
- [3] F.J. Almgren, Some interior regularity theorems for minimal surfaces and an extension of Bernstein's theorem, Annals of Mathematics. Second Series 84 (1966), 277-292.
- [4] L. Ambrosio, X. Cabré, Entire solutions of semilinear elliptic equations in R3 and a conjecture of De Giorgi, Journal Amer. Math. Soc. 13 (2000), 725-739.
- [5] S. Baraket, F. Pacard, Construction of singular limits for a semilinear elliptic equation in dimension 2. Calc. Var. Partial Differential Equations 6 (1998),no. 1, 1-38.
- [6] S.N. Bernstein, Sur une theoreme de geometrie et ses applications aux equations derivees partielles du type elliptique, Comm. Soc. Math. Kharkov 15 (1915-1917) 38-45.
- [7] E. BOMBIERI, E. DE GIORGI, E. GIUSTI, Minimal cones and the Bernstein problem. Invent. Math. 7 1969 243-268.
- [8] C.J. COSTA, Example of a complete minimal immersions in R3 of genus one and three embedded ends, Bol. Soc. Bras. Mat. 15(1-2)(1984), 47-54.
- [9] E. N. DANCER, Stable and finite Morse index solutions on Rn or on bounded domains with small diffusion. Trans. Amer. Math. Soc. 357 (2005), no. 3, 1225-1243.
- [10] J. DÁVILA, L.F. LÓPEZ, Regular solutions to a supercritical elliptic problem in exterior domains. J. Differential Equations 255 (2013), no. 4, 701-727.
- [11] E. DE GIORGI, Una estensione del teorema di Bernstein, Ann. Scuola Norm. Sup. Pisa (3) 19 (1965) 79-85.
- [12] E. DE GIORGI, Convergence problems for functionals and operators, Proc. Int. Meeting on Recent Methods in Nonlinear Analysis (Rome, 1978), 131-188, Pitagora, Bologna (1979).
- [13] M. Del Pino, M. Kowalczyk, J, Wei, Entire solutions of the Allen-Cahn equation and complete embedded minimal surfaces of finite total curvature. J. Diff. Geometry vol. 93 (2013).
- [14] M. DEL PINO, M. KOWALCZYK, W. JUNCHENG, Traveling waves with multiple and nonconvex fronts for a bistable semilinear parabolic equation. Comm. Pure Appl. Math. 66 (2013), no. 4, 481-547. 35K91 (35C07)
- [15] M. DEL PINO, M. KOWALCZYK, F. PACARD, J.C. WEI, Multiple end solutions to the Allen-Cahn equation in ℝ². J. Funct. Analysis 258(2010), no.2, 458-503.
- [16] M. Del Pino, M. Kowalczyk, J. Wei, On De Giorgi Conjecture in Dimensions $N \geq 9$. Annals of Mathematics 174 (2011), no.3, 1485-1569.
- [17] M. DEL PINO, M. KOWALCZYK, J.C. WEI, J. YANG, Interface foliations of a positively curved manifold near a closed geodesic. Geom. Funct. Anal. 20 (2010), no. 4, 918-957.
- [18] S. Fakhi, F. Pacard, Existence result for minimal hypersurfaces with a prescribed finite number of planar ends. Manuscripta Math. 103 (2000), no. 4, 465-512.
- [19] W.H. Fleming, On the oriented Plateau problem, Rendiconti del Circolo Matematico di Palermo. Serie II 11 (1962), 69-90.
- [20] A. Floer, A. Weinstein, Nonspreading wave packets for the cubic Schrödinger equation with a bounded potential. J.Funct. Anal. 69 (1986), no. 3, 397-408.
- [21] N. GHOUSSOUB, C. Gui, On a conjecture of De Giorgi and some related problems, Math. Ann. 311 (1998),481-491.
- [22] P. HARTMAN, Ordinary Differential Equations: An introduction. John Wiley & Sons Inc.New York, 1964.
- [23] D. HOFFMAN AND W.H. MEEKS III, A complete embedded minimal surface in R3 with genus one and three ends, J. Diff. Geom. 21 (1985), 109-127.
- [24] D. HOFFMAN AND W.H. MEEKS III, Embedded minimal surfaces of finite topology, Ann. Math. 131 (1990), 1-34.
- [25] N. KAPOULEAS, Complete embedded minimal surfaces of finite total curvature. J. Differential Geom. 45 (1997), 95-169.
- [26] L. Modica, Γ -convergence to minimal surfaces problem and global solutions of $\Delta u = 2(u3 u)$. Proceedings of the International Meeting on Recent Methods in Nonlinear Analysis (Rome, 1978), Pitagora, Bologna, 1979, pp. 223-244. MR 0533169. Zbl 0408.49041.
- [27] J. Pérez, A. Ros, The space of properly embedded minimal surfaces with finite total curvature. Indiana Univ. Math. J. 45 (1996), no. 1, 177-204.
- [28] L-F. TAM, D. ZHOU, Stability properties for the higher dimensional catenoid in ℝⁿ⁺¹. Proc. Amer. Math. Soc. 137(2009), no.10, 3451-3461.
- [29] O. SAVIN, Regularity of at level sets in phase transitions, Annals of Mathematics, 169 (2009), 41-78.

- [30] F. Pacard, J.C. Wei, Stable solutions of the Allen-Cahn equation in dimension 8 and minimal cones. J. Funct. Anal. 264 (2013), no. 5, 1131-1167. 35J91 (35B09)
 - O. AGUDELO NTIS, DEPARTMENT OF MATHEMATICS, ZAPADOCESKA UNIVERZITA V PLZNI, PLZEN, CZECH REPUBLIC. $E\text{-}mail\ address:}$ oiagudel@ntis.zcu.cz
- M. Del Pino Departamento de Ingeniería Matemática and CMM, Universidad de Chile, Casilla 170 Correo 3, Santiago, Chile.

 $E\text{-}mail\ address: \ \mathtt{delpino@dim.uchile.cl}$

J. Wei - Department of Mathematics, University of British Columbia, Vancouver BC V6T 1Z2, Canada and Department of Mathematics, Chinese University of Hong Kong, Shatin, NT, Hong Kong *E-mail address*: wei@math.cuhk.hk